Contents lists available at ScienceDirect

Journal of Banking & Finance

journal homepage: www.elsevier.com/locate/jbf



Information asymmetry and the value of cash

Wolfgang Drobetz^{a,*}, Matthias C. Grüninger^b, Simone Hirschvogl^c

^a Institute of Finance, University of Hamburg, 20146 Hamburg, Germany

^b Department of Finance, University of Basel, 4002 Basel, Switzerland

^c Corporate Finance, KPMG LPP (UK), EC4Y 8BB London, United Kingdom

ARTICLE INFO

Article history: Received 8 March 2009 Accepted 2 February 2010 Available online 6 February 2010

JEL classification: G32

Keywords: Cash holdings Value of cash Information asymmetry Analysts' forecasts

ABSTRACT

This study investigates the market value of corporate cash holdings in connection with firm-specific and time-varying information asymmetry. Analyzing a large international sample, we test two opposing hypotheses. According to the pecking order theory, adverse selection problems make external financing costly and imply a higher market value of a marginal dollar of cash in states with higher information asymmetry. In contrast, the free cash flow theory predicts that excessive cash holdings bundled with higher information asymmetry generate moral hazard problems and lead to a lower market value of a marginal dollar of cash. We use the dispersion of analysts' earnings per share forecasts as our main measure of firm-specific and time-varying information asymmetry. Extending the valuation regressions of Fama and French [Fama, E.F., French, K.R., 1998. Taxes, financing decisions, and firm value. Journal of Finance 53, 819–843], our results support the free cash flow theory and indicate that the value of corporate cash holdings is lower in states with a higher degree of information asymmetry.

© 2010 Elsevier B.V. All rights reserved.

1. Introduction

J.P. Morgan economists calculated that savings by corporations in rich countries increased by more than \$1 trillion from 2000 to 2004.¹ Compared to the last 40 years, firms never hoarded so much cash as they did during this recent time period. A natural question is why firms accumulate such enormous amounts of liquidity. The standard textbook model suggests that cash holdings are irrelevant and cannot affect firm value. In perfect (frictionless) capital markets, external finance can always be obtained at fair terms. Looking at the corporate landscape, however, this cash irrelevancy is not supported. For example, the US software giant Microsoft presented a cash position amounting to \$60.6 billion in its 2004 annual report. After growing investor pressure, in July 2004 Microsoft announced to pay a one-time dividend of \$32 billion and to buy back up to \$30 billion of the company's stock over the next 4 years. Upon the arrival of that news, Microsoft's stock price rose by 5.7% in the after-trading, indicating that cash should by no means be regarded as irrelevant in investors' eyes.²

In order to explain corporate cash holdings, it is necessary to relax the assumptions of frictionless capital markets. First, if transaction costs are incorporated into the model, the irrelevancy proposition of cash no longer holds and an optimal cash balance exists. Second, if information asymmetry is taken into account, adverse selection and moral hazard problems arise. Myers and Majluf (1984) model the adverse selection problem in financing decisions and consider the role of cash holdings in the presence of information asymmetry. Adverse selection induces managers to abstain from raising external capital because they are not willing to issue undervalued securities. A cash buffer may prevent managers from being forced to pass up positive net present value investment projects. In contrast, Jensen (1986) analyzes the moral hazard problem and emphasizes the agency costs of free cash flow. Instead of paying out the free cash flow to shareholders, managers tend to waste these funds on inefficient investments or on their own pet projects (empire building).

Corporate cash holdings and information asymmetry are strongly interrelated. This is the novel path that our study takes and how it contributes to the existing literature. Specifically, we measure the marginal value of cash holdings in the presence of firm-specific and time-varying information asymmetry. Previous studies also investigate the value consequences of corporate cash holdings, but they put their emphasis on corporate governance issues rather than on information asymmetry. These studies document that a weak corporate governance regime has detrimental effects on the value of cash (Dittmar et al., 2003; Pinkowitz et al., 2006; Dittmar and Mahrt-Smith, 2007). In this study, we analyze firm-specific and time-varying information asymmetry and its impact on the market value of cash. We test whether in states with a

^{*} Corresponding author. Tel.: +49 40 42838 5506; fax: +49 40 42838 4627.

E-mail addresses: wolfgang.drobetz@wiso.uni-hamburg.de (W. Drobetz), matthias.grueninger@alumnibasel.ch (M.C. Grüninger), simone.hirschvogl@kpmg.co.uk (S. Hirschvogl).

J.P. Morgan Research: Corporates are driving the global saving glut, June 24, 2005.
 The Wall Street Journal, Microsoft to dole out its cash hoard, July 21, 2004, p. A.1.

^{0378-4266/\$ -} see front matter © 2010 Elsevier B.V. All rights reserved. doi:10.1016/j.jbankfin.2010.02.002

higher degree of information asymmetry cash holdings contribute more or less to firm value than in states with a lower degree of information asymmetry. On the one hand, a positive relationship supports Myers and Majluf's (1984) hypothesis that external finance is costly and cash provides a valuable buffer. On the other hand, a negative relationship is consistent with Jensen's (1986) notion that increased managerial discretion induces managers to squander corporate liquidity. We test these two opposing hypotheses and investigate which effect outweighs the other. Our sample contains more than 8500 firms from 45 countries over the period from 1995 to 2005. The dispersion of analysts' earnings forecasts serves as our main proxy for information asymmetry. Extending the Fama-French (1998) valuation regressions, we include the cash ratio and compute its impact on firm valuation in connection with firm-specific and time-varying information asymmetry. Methodologically, we use fixed effects regressions and the Fama-MacBeth procedure.

Without considering information asymmetry, our results indicate that the value shareholders place on the marginal unit of cash is around one dollar, on average, depending on the estimation methodology. Most important, the marginal value of cash decreases with increasing severity of information asymmetry. This evidence supports Jensen's (1986) free cash flow theory, i.e., the costs from holding cash (creating moral hazard problems) outweigh the benefits (avoiding costly external finance). Incorporating a measure of excess cash in the valuation regressions instead of the actual cash ratio does not change our results qualitatively. In order to further distinguish between the two opposing hypotheses, we split the sample according to measures for the quality of corporate governance and the degree of financial constraints. We find that the value of cash is higher if corporate governance and investor protection are better, which reinforces the free cash flow hypothesis. In contrast, our results for the sample sorts based on financial constraints do not allow unambiguous conclusions.

We are unable to support the hypothesis that financial slack is valuable, as predicted by the pecking order theory. Our findings indicate that it may not be in the shareholders' interest that firms hoard liquidity due to problems induced by information asymmetry, and hence the precautionary motive to hold cash appears questionable. However, they do not generally contradict the pecking order theory. In particular, our results do not suggest that firms should not use internal funds in the first place before external funds are raised. We rather argue that it may not be optimal for firms to accumulate cash with the intention to avoid (costly) external finance in future states when information asymmetry is high.

The remainder of this paper is structured as follows. Section 2 introduces the theoretical background, presents our hypotheses, and reviews the related literature. Section 3 describes the data and explains our empirical methodology. Section 4 reports our main empirical results and a number of robustness tests. Finally, Section 5 provides concluding remarks and suggestions for further research.

2. Theoretical background, hypotheses, and related literature

2.1. Theoretical background and hypotheses

According to the pecking order theory (Myers, 1984; Myers and Majluf, 1984), firms prefer internal to external finance. This theory is based on the assumption that corporate insiders are better informed than shareholders. Managers may be forced to forgo profitable projects if internal funds are not sufficient to finance the optimal investment program and information asymmetry is prohibitive. In this situation, financial slack (cash) is valuable, and the only opportunity to issue equity without loss of market value occurs if information asymmetry is nonexistent or small. This idea captures the notion of time-varying adverse selection costs (Korajczyk et al., 1992; Viswanath, 1993). It can be optimal for firms to deviate from the strict pecking order and to finance a new investment project with fresh equity even if there are other financing options available. Specifically, there are states in which firms are not restricted to raise external capital, and there are other states in which the costs of external finance are excessive. In those states when external finance is prohibitively expensive, financial slack is valuable, and an additional dollar of cash will have a higher market value. This reasoning results in our first hypothesis:

Hypothesis 1. In states with a higher degree of information asymmetry cash is *more* valuable for a firm than in states with a lower degree of information asymmetry.

The opposite relationship could be expected based on Jensen's (1986) free cash flow theory. More internal funds allow managers to elude control of the capital market. In this case, they do not need shareholders' approval and are free to decide on investments according to their own discretion. Managers are reluctant to pay out funds, and they have an incentive to invest even when there are no positive net present value projects available. With increasing managerial discretion to misuse funds for value-destroying projects when cash reserves are high, corporate governance mechanisms, e.g., the market for corporate control, are supposed to limit self-serving behavior. However, the higher the degree of information asymmetry, the more difficult it becomes for firm outsiders to distinguish between value-destroying and value-increasing investments. Specifically, shareholders may be unable to determine whether high cash reserves are close to the amount required for the firm to operate or whether they are the result of managerial risk aversion (Fama and Jensen, 1983). This reasoning results in our second hypothesis:

Hypothesis 2. In states with a higher degree of information asymmetry cash is *less* valuable for a firm than in states with a lower degree of information asymmetry.

Our hypotheses contain opposing expectations concerning the influence of information asymmetry on the value of cash holdings. The main challenge in our empirical tests is to disentangle the effects of these two conflicting hypotheses. If no relationship is detected, we cannot rule out that both effects are at work and cancel each other out. Even if a relationship can be detected, we still cannot rule out that the opposite effect also exists, albeit to a lesser degree. Although we are ultimately interested in the overall (net) effect, we attempt to disentangle the two effects by splitting our sample into different subgroups. Specifically, Hypothesis 1 is related to the access to external finance. Splitting the sample according to the severity of financial constraints, one expects that in the subsample of constrained firms the value of cash is higher with more pronounced information asymmetry. This finding would support Hypothesis 1, regardless of the overall (net) effect. In contrast, Hypothesis 2 is more relevant for firms with weaker corporate governance structures. Splitting the sample according to the quality of corporate governance, the value of cash in combination with a high degree of information asymmetry should be lower in the subsample of firms with weaker corporate governance practices. This finding would support Hypothesis 2, regardless of the overall (net) effect.

2.2. Related literature

This section reviews research findings that are related to our hypotheses. One strand of the literature provides evidence for a dynamic version of the pecking order theory (related to Hypothesis

1). Bharat et al. (2008) and Autore and Kovacs (2010) document that firms prefer to access financial markets and issue equity when the level of information asymmetry is low. Similarly, Krishnaswami and Yaman (2007) study windows of opportunity in debt issues. In the spirit of Korajczyk et al. (1992), Choe et al. (1993) and Viswanath (1993), these studies document support for a time-varying adverse selection explanation of firms' financing decisions. Based on their findings, one would expect that cash is more important for firms and has a higher market value in states when information asymmetry is more pronounced. In contrast, Fama and French (2005) and Leary and Roberts (2007) report that the pecking order theory is not able to explain firms' financing decisions even in states when information asymmetry is high. D'Mello et al. (2008) analyze the initial cash allocation decision in spin-off firms. They document that spin-off firms with high information asymmetry hold more cash in order to reduce their dependence on costly external finance. However, an analysis of the excess cash ratio indicates that firms, on average, hold less cash than is suggested by the trade-off theory. Observing that the excess cash ratio is positively related to the cash flow in the current year, D'Mello et al. (2008) attribute this conservatism in cash holdings to pecking order effects.

Another strand of the literature that is important for our analysis tests the free cash flow hypothesis (related to Hypothesis 2). For example, Nohel and Tarhan (1998) investigate the impact of share repurchases on operating performance. Their findings reveal that operating performance improves after share repurchases, but only for firms with low growth opportunities. This improved performance does not result from higher growth opportunities but from the more efficient employment of assets. Nohel and Tarhan (1998) argue that their findings support the free cash flow hypothesis. D'Mello and Miranda (2010) provide more direct evidence on the agency costs of managerial discretion in connection with corporate cash holdings. They document that the introduction of debt in unlevered firms leads to a sharp decline in cash ratios, and this relation is stronger for firms with poor investment opportunities. Using international data, Dittmar et al. (2003) report that firms in countries with a low level of investor protection hold double the amount of cash compared to firms in countries with a high level of shareholder rights. Their results are even more pronounced when they control for the capital market development. Pinkowitz and Williamson (2004) and Pinkowitz et al. (2006) also focus on the influence of country-level investor protection on the value of cash holdings. Their results reveal that cash is worth less in countries where minority shareholder rights are poorer. Similarly, Fresard and Salva (forthcoming) document that the value investors attach to excess cash holdings is higher for foreign firms with US cross-listings than their domestic peers.

Harford et al. (2008) report that US firms with poor corporate governance tend to hold lower cash and contribute this result to the observation that they invest less internally but engage more frequently in acquisitions, often using cash as their method of payment.³ These acquisitions, as well as the lower internal investment, destroy firm value through reduced future profitability. At first, these findings for US firms seem to be inconsistent with international studies that document decreasing cash balances as shareholder rights increase. As an explanation, Harford et al. (2008) suggest that true entrenchment requires low shareholder rights. The effect of country-level granting and enforcing of shareholder

rights dominates the effect of firm-level variation in the control of agency conflicts. The level of entrenchment found in environments with poor shareholder protection is generally higher than that in the average entrenched firm in countries with higher shareholder protection. While managers can hoard cash with impunity in the former case, they are wary of opposing shareholder agitation in the latter case if entrenchment goes too far. Supporting this notion, Kalcheva and Lins (2007) document that firms with weak corporate governance structures at the firm-level hold more cash, and this effect becomes stronger for firms in countries with poor investor protection. Overall, these studies support the free cash flow hypothesis.⁴ Poor protection of investor rights enables managers to dissipate cash for their own ends.

Finally, Lundstrum (2003) explicitly focuses on information asymmetry and corporate cash holdings. He investigates whether the benefits from accessing internal capital markets to avoid selling underpriced securities outweigh the agency costs from managerial money squandering created by the availability of cash. His results corroborate the free cash flow theory. Although access to internal capital markets can sometimes exert a positive effect on firm value, this effect only arises for firms with a low level of information asymmetry.

3. Data and empirical methodology

The starting point of our analysis is the valuation model of Fama and French (1998). They investigate how firm value is related to dividends and corporate debt. Pinkowitz and Williamson (2004), Pinkowitz et al. (2006) and Dittmar and Mahrt-Smith (2007) suggest modified versions of this valuation model to estimate the market value of a dollar of cash. We further extend their framework and test the impact of firm-specific and time-varying information asymmetry on the market value of cash holdings. Section 3.1.1 starts with a description of our international sample. Section 3.1.2 introduces different proxies for time-varying and firm-specific information asymmetry, and Section 3.1.3 suggests sample splits that are based on financial constraints and corporate governance structures. Finally, Section 3.2 explains our empirical methodology and introduces all other model variables.

3.1. Data description

3.1.1. The sample

Our data set covers the period from 1995 to 2005. We include all firms from the different countries for which I/B/E/S provides analysts' forecasts and for which we can retrieve company data from Worldscope. Given that our main proxy for information asymmetry is based on the standard deviation of analysts' earnings per share forecasts (analysts' forecasts dispersion), this measure can only be computed when the forecasts are at least based on two analysts. A firm is omitted from our sample if this dispersion measure cannot be calculated for at least one sample year, hence, if this firm is not covered by at least two analysts in at least one sample year. We use yearly data because for most countries quarterly accounting data are not available. Given the specific nature of their business, financial firms and utilities are omitted from the sample. Moreover, firms whose fiscal year does not end with the calendar year are excluded. This data restriction avoids that our financial data, especially the earnings estimates that are used to compute our information asymmetry measures, refer to different

³ Because large cash holdings may lead to shareholder agitation in countries with high investor protection, managers prefer to convert the cash into real assets relatively quickly. Even if these transactions may be value-destroying, managers can execute them successfully as long as the costs are within the bounds of removing management.

⁴ In contrast, examining a US sample, Bates et al. (2009) report that the recent increase in cash holdings can be explained by changing firm characteristics (and hence the precautionary demand for cash) rather than agency problems.

Table 1					
Observations	per	country	and	index	values.

Country	N Model 1	N Model 2	Anti-director rights index	Corruption index	Rule of law index	Common law	Civil law	Stock/GDP	Bond/GDP
Panel A: Developed	d markets								
Australia	355	292	4	2.00	1.89	1	0	1.04	0.28
Austria	331	283	2	1.88	1.94	0	1	0.17	0.35
Belgium	464	398	0	1.32	1.53	0	1	0.81	0.46
Canada	1699	1110	5	2.25	1.87	1	0	1.16	0.22
Denmark	517	82	2	2.31	1.87	0	1	0.68	1.03
Finland	743	658	3	2.49	2.02	0	1	2.70	0.24
France	2400	2084	3	1.41	1.36	0	1	1.13	0.40
Germany	2493	2119	1	1.67	1.84	0	1	0.73	0.62
Greece	758	176	2	0.84	0.66	0	1	1.42	0.00
Hong Kong	1061	89	5	1.43	1.44	1	0	3.76	0.18
Ireland	217	211	4	1.50	1.71	1	0	0.80	0.08
Italy	961	842	1	0.79	0.88	0	1	0.70	0.33
lapan	1103	0	4	1.28	1.66	0	1	0.82	0.47
The Netherlands	1078	950	2	2.30	1.89	0	1	1.81	0.47
New Zealand	45	34	4	2.31	1.89	1	0	0.45	
Norway	657	91	4	2.07	1.90	0	1	0.39	0.20
Portugal	227	211	3	1 37	1.07	0	1	0.60	0.25
Singapore	840	654	4	2.44	1.91	1	0	1 93	0.18
Snain	678	586	4	1.62	1.29	0	1	0.84	0.15
Sweden	1146	116	3	2.43	1.87	0	1	1 47	0.43
Switzerland	940	863	2	2.13	2.11	0	1	3.03	0.43
United Kingdom	3003	2686	5	2.10	1.80	1	0	1 93	0.20
United States	14,419	12,234	5	1.73	1.79	1	0	1.64	1.02
Panel R. Emerging	markets								
Argentina	156	145	4	-0.40	0.07	0	1	0.44	0.05
Brazil	540	382	3	_0.01	_0.21	0	1	0.38	0.09
Chile	478	78	5	1 50	1 23	0	1	0.86	0.05
China	917	,0		_0.38	_0.42	0	1	0.00	0.09
Colombia	55	0	3	-0.50	_0.73	0	1	0.42	0.00
Czech Republic	65	0		0.39	0.51	0	1	0.15	0.00
Hungary	109	0		0.71	0.77	0	1	0.21	0.07
India	146	0	5	_0.31	0.15	1	0	0.37	0.02
Indonesia	636	0	2	-1.05	_1.03	0	1	0.28	0.00
Israel	186	89	3	1.03	0.96	1	0	0.56	
Malavsia	1012	362	4	0.21	0.39	1	0	1 46	0.49
Mexico	655	181	1	-0.49	-0.45	0	1	0.24	0.02
Pakistan	55	0	5	-0.94	-0.75	1	0	0.09	
Peru	107	77	3	-0.16	-0.60	0	1	0.03	0.04
Philippines	286	0	3	_0.53	-0.55	0	1	0.66	0.00
Poland	238	77		0.48	0.54	0	1	0.00	0.00
Russia	67	0		-1.04	-0.99	0	1	0.22	
South Africa	191	52	5	0.49	0.15	1	0	1 77	0.09
South Korea	2534	0	2	0.33	0.52	0	1	0.56	0.40
Taiwan	22334	0	3	0.63	0.76	0	1	1.02	0.26
Thailand	1088	0	2	-0.37	0.30	1	0	0.36	0.12
Turkey	297	265	2	_0.36	_0.07	0	1	0.46	0.12
rancy	231	205	2	0.50	0.07	0	•	0.40	

This table shows the number of firm-year observations (*N*) for the countries that are included in the main valuation regression (model 1) and the alternative specification that incorporates a target cash level (model 2). The table also presents the values of the indices for the year 2000 that are used to split the sample into subgroups based on country characteristics. The anti-director rights index measures the protection of shareholder rights, where higher index values indicate better protection of minority shareholder rights. The rule of law index and the corruption index both measure the quality of institutions that support the rights of investors. Firms in countries with lower or even negative index values operate under weaker corporate governance structures, and it is more difficult for investors to make use of their formal rights. A dummy variable indicates a country's law tradition (common law or civil law). All these measures are related to a country's overall quality of corporate governance. In order to measure financial constraints on a country-level, stock market capitalization to GDP (stock/GDP) is defined as the ratio of the market value of listed shares in a country to its gross domestic product. Private bond market capitalization to GDP (bond/GDP) is the ratio of a country's private domestic debt securities to its gross domestic product. A dot indicates that the value is not defined.

time periods.⁵ To reduce the impact of outliers, we trim all variables at the 1% and the 99% tails. Finally, we exclude countries with fewer than 30 firm-year observations. In the most basic specification, our sample consists of 8661 firms with 48,240 firm-year observations from 45 countries. Table 1 contains a list of the countries contained in our sample together with descriptive statistics of the country-level variables. Panel B of Table 2 shows descriptive statistics of the firm-level variables that are used in our empirical models.

3.1.2. Measures of information asymmetry

In order to test the relationship between the value of cash and information asymmetry, a reliable proxy for information asymmetry is required. Based on Krishnaswami and Subramaniam (1999), we choose the dispersion of analysts' forecasts as our main proxy for information asymmetry.⁶ This variable measures the standard deviation of earnings per share forecasts across analysts that cover

 $^{^5}$ This data restriction eliminates only about 1% of all firm-year observations from the sample. Most firms that drop out are from Japan, where the most popular fiscal year end is March 31. Our results are robust when these firms remain in the base sample.

⁶ Krishnaswami and Subramaniam (1999) also discuss other proxy variables for information asymmetry, e.g., the volatility of abnormal returns around earnings announcements and the volatility of daily stock returns. While return volatility around earnings announcements is not a feasible measure of information asymmetry in a cross-country study, daily stock return volatility does not allow to distinguish between risk in a broader sense and the effect of information asymmetry.

Descriptive statistics.

Fund. A. Dispersion (MSPM) on country-level Germany 1614 0.042 0.273 0.154 0.676 0.215 Tray 177 0.069 0.241 0.165 0.525 0.238 United Kingdom 2.267 0.023 0.113 0.052 0.364 0.225 Canada 1302 0.037 0.152 0.364 0.225 Canada 1302 0.037 0.179 0.091 0.446 0.252 All developed markets 7705 0.044 0.214 0.129 0.477 0.252 All developed markets 7017 0.707 0.228 0.192 0.627 0.285 All developed markets 7017 0.707 0.285 0.192 0.627 0.285 All constrice 707 0.044 0.214 0.063 1.060 0.371 Herealter measures for information asymmetry ER ER 8.033 1.037 0.1400 18.000 0.325 V1 48.240 -0.042		Ν	p10	Mean	Median	p90	S.D.					
Forecast dispersion (JSMP) on country-level Germany 1614 0.048 0.273 0.154 0.665 0.213 Italy 7.57 0.069 0.241 0.165 0.525 0.238 United Kingdom 2.267 0.023 0.129 0.060 0.266 0.214 United Stares 11.718 0.013 0.137 0.052 0.364 0.225 Ganada 1302 0.037 0.275 0.156 0.016 0.237 Other developed markets 7.706 0.044 0.214 0.129 0.477 0.252 Interging markets 7.017 0.070 0.285 0.192 0.627 0.282 All developed markets 7.706 0.044 0.214 0.407 0.252 All developed markets 7.706 0.044 0.214 0.477 0.252 All developed markets 7.059 0.000 0.344 0.180 0.827 0.251 All courbins 0.592 0.000 1.40	Panel A: Dispersion measures											
	Forecast dispersion (DISPM) on country-level											
France 1837 0.042 0.202 0.115 0.450 0.2255 Italy 757 0.069 0.241 0.165 0.5255 0.238 United Kingdom 2267 0.023 0.129 0.060 0.266 0.224 Lunited States 1.718 0.013 0.137 0.055 0.056 0.237 Canada 1302 0.037 0.275 0.156 0.503 0.237 Other developed markets 27,859 0.019 0.179 0.919 0.446 0.250 All cocoloped markets 7,755 0.002 0.228 0.192 0.627 0.283 All cocoloped markets 7,857 0.002 0.244 0.003 1.060 0.571 All cocoloped markets 7,875 0.000 0.344 0.003 1.060 3.290 Parel B: Firm-level variables used in the main valuation regression Vr. 4.8240 -0.354 0.007 0.065 0.065 0.066 Vr. 4.8240 -0.035 <td>Germany</td> <td>1614</td> <td>0.048</td> <td>0.273</td> <td>0.154</td> <td>0.676</td> <td>0.310</td>	Germany	1614	0.048	0.273	0.154	0.676	0.310					
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	France	1837	0.042	0.202	0.115	0.450	0.255					
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	Italy	757	0.069	0.241	0.165	0.525	0.238					
	United Kingdom	2267	0.023	0.129	0.060	0.266	0.214					
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	United States	11,718	0.013	0.137	0.052	0.364	0.225					
$\begin{tabular}{ c c c c c c c c c c c c c c c c c c c$	Canada	1302	0.037	0.275	0.156	0.716	0.316					
$\begin{tabular}{ c c c c c c c c c c c c c c c c c c c$	Japan	658	0.033	0.197	0.116	0.503	0.237					
All developed markets27,8590.0190.1790.0910.4460.250Emerging markets70170.0700.2850.1920.6270.281All countries34,8760.0220.2020.1120.4870.261Alternative measures for information asymmetry $=$ ERRORF29,9820.0000.3440.0831.0609.971IA-INDEX29,2289.00013,60014,00018,0003.290Panel E: Firm-level variables used in the main valuation regression $=$ $=$ V.48,240-0.3940.1590.0420.7950.885Er48,240-0.0540.0060.0070.0650.066det_i.148,240-0.0550.0100.0080.0760.076dWr.48,240-0.1230.0920.0450.3410.255dWr.48,240-0.1230.0920.0450.3410.255dWr.48,240-0.0120.0620.0520.2800.185dWr.48,240-0.0130.0920.0450.3410.255dRDr.48,240-0.0100.0010.0000.0050.012dRDr.48,240-0.0100.0010.0000.0050.012dWr.48,240-0.0100.0010.0000.0050.012dth.48,240-0.0010.0010.0000.0100.010dth.48,240-0.0100.0020.0000.01	Other developed markets	7706	0.044	0.214	0.129	0.477	0.252					
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	All developed markets	27,859	0.019	0.179	0.091	0.446	0.250					
All countries34,8760.0220.2020.1120.4870.261Alternative measures for information asymmetryERRORF29,9820.0000.3440.0831.0600.971IA-INDEX29,2289.00013.60014.00018.0003.290Vr.48,2400.0541.2500.9452.3101.010 V_t 48,240-0.03940.1590.0420.7950.885 E_t 48,240-0.0420.0520.0600.1540.164 dE_t 48,240-0.0550.0100.0080.0760.076 dN_k 48,240-0.1230.0620.0520.2800.185 dN_k 48,240-0.0130.0010.0000.0050.012 dR_k 48,240-0.0010.0010.0000.0050.012 dR_k 48,240-0.0010.0010.0000.0060.012 dR_k 48,240-0.0010.0010.0000.0100.011 dR_k 48,240-0.0010.0010.0000.0110.011 dR_k 48,240-0.0050.0010.0000.0110.011 dR_k 48,240-0.0080.0010.0000.0160.020 dR_k 48,240-0.0050.0020.0620.0220.0110.011 dR_k 48,240-0.0050.0010.0000.0110.011 dR_k 48,240-0.0080.001 <td>Emerging markets</td> <td>7017</td> <td>0.070</td> <td>0.285</td> <td>0.192</td> <td>0.627</td> <td>0.285</td>	Emerging markets	7017	0.070	0.285	0.192	0.627	0.285					
Alternative measures for information asymmetryERRORF29.9820.0000.3440.0831.0600.9711Panel B: Firm-level variables used in the main valuation regressionV48.2400.5041.2500.9452.3101.010 V_{t+1} 48.240-0.03440.1590.0420.7950.885 E_t 48.240-0.0420.0520.0660.1540.101 dE_t 48.240-0.0540.0060.0070.0650.066 dE_t 48.240-0.1230.0920.0450.3410.255 dE_t 48.240-0.1230.0920.0450.3410.255 dR_{t+1} 48.240-0.0130.0090.0050.012 dR_t 48.240-0.0100.0010.0000.0050.012 dR_t 48.240-0.0010.0010.0000.0050.012 dR_t 48.240-0.0080.0010.0000.0060.012 dR_{t+1} 48.240-0.0080.0010.0000.0110.011 dR_t 48.240-0.0080.0010.0000.0160.0430.019 dt_{t+1} 48.240-0.0080.0010.0000.0160.0220.0220.0230.016 dR_t 48.240-0.0050.0120.0000.0160.0230.0230.0230.0230.0230.0230.0230.0230.0230.0230.0230.0230.0330.0000.0	All countries	34.876	0.022	0.202	0.112	0.487	0.261					
Alternative measures for information symmetryERRORF29.9820.0000.3440.0831.0600.971IA-INDEX29.2289.00013.60014.00018.0003.290Panel E: Firm-level variables used in the main valuation regressionVi48.240-0.5041.2500.9452.3101.010dVi+148.240-0.0340.01590.0420.7950.885Ei48.240-0.0340.0060.0070.0650.066dEi,148.240-0.1200.0620.0520.2800.1340.255dNA_148.240-0.1230.0920.0440.3410.255RD,48.240-0.0100.0010.0000.0050.012dRD,48.240-0.0010.0010.0000.0060.012dRD,48.240-0.0010.0010.0000.0060.012dRD,48.240-0.0080.0010.0000.0100.010dIt,48.240-0.0080.0010.0000.0110.011dIt,48.240-0.0080.0010.0000.0160.023dL,148.240-0.0050.0020.0000.0160.023dR D,48.240-0.0080.0010.0000.0100.010dIt,48.240-0.0080.0010.0000.0160.023dIt,48.240-0.0050.0020.0020.0240.027	Alt											
ERROR 29,982 0.000 0.344 0.083 1.060 0.971 IA-INDEX 29,228 9,000 13,600 14,000 18,000 3,290 Panel B: Firm-level variables used in the main valuation regression - 48,240 0.504 1,250 0.945 2,310 1.010 dV_{t+1} 48,240 -0.042 0.052 0.060 0.154 0.104 E_t 48,240 -0.055 0.010 0.008 0.076 0.070 dK_{t+1} 48,240 -0.123 0.062 0.052 0.280 0.185 dNA_{t+1} 48,240 -0.123 0.062 0.052 0.280 0.012 dRD_{t+1} 48,240 -0.001 0.001 0.000 0.053 0.042 dRD_{t+1} 48,240 -0.001 0.001 0.000 0.005 0.012 dRD_{t+1} 48,240 -0.008 0.001 0.000 0.011 0.011 dRD_{t+1} 48,240 -0.008 0.001 </td <td></td> <td></td> <td>0.000</td> <td>0.244</td> <td>0.002</td> <td>1.000</td> <td>0.071</td>			0.000	0.244	0.002	1.000	0.071					
μ -/NDEA 29,228 9,000 13,000 14,000 18,000 3,230 Panel B: Firm-level variables used in the main valuation regression	ERKURF	29,982	0.000	0.344	0.083	1.060	0.971					
Panel B: Firm-level variables used in the main valuation regression V_t 48,240 0.504 1.250 0.945 2.310 1.010 dV_{t+1} 48,240 -0.334 0.159 0.042 0.755 0.885 E_t 48,240 -0.055 0.010 0.006 0.007 0.0655 0.066 dE_{t+1} 48,240 -0.120 0.062 0.052 0.280 0.183 dN_{t+1} 48,240 -0.123 0.092 0.045 0.341 0.255 RD_t 48,240 -0.012 0.002 0.000 0.005 0.012 dRD_t 48,240 -0.001 0.001 0.000 0.005 0.012 dRD_t 48,240 -0.001 0.001 0.000 0.005 0.012 l_t 48,240 -0.008 0.001 0.000 0.010 0.001 dRD_t 48,240 -0.008 0.001 0.000 0.011 0.011 D_t 48,240	IA-INDEX	29,228	9.000	13.600	14.000	18.000	3.290					
V_t 48,2400.5041.2500.9452.3101.010 dV_{t+1} 48,240-0.3940.1590.0420.7950.885 E_t 48,240-0.0540.0660.0070.0650.066 dE_t 48,240-0.0550.0100.0080.0760.070 dN_t 48,240-0.1200.0620.5520.2800.185 dN_{t+1} 48,240-0.1230.0920.0450.3410.255 dN_{t+1} 48,240-0.0010.0010.0000.0050.012 dR_{t+1} 48,240-0.0010.0010.0000.0050.012 dR_{t+1} 48,240-0.0010.0010.0000.0060.012 dR_{t+1} 48,240-0.0010.0010.0000.0160.012 dR_{t+1} 48,240-0.0080.0010.0000.0100.010 dR_{t+1} 48,240-0.0080.0010.0000.0110.011 dL_t 48,240-0.0080.0010.0000.0160.023 dL_t 48,240-0.0080.0010.0000.0160.023 dL_t 48,240-0.0100.0020.0000.0160.023 dL_t 48,240-0.0100.0020.0000.0160.023 dL_t 48,240-0.0110.0020.0000.0160.023 dL_t 48,240-0.0110.0050.0020.0910.034 dL_t 48,24	Panel B: Firm-level variables used in	the main valuation regre	ssion									
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	V _t	48,240	0.504	1.250	0.945	2.310	1.010					
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	dV_{t+1}	48.240	-0.394	0.159	0.042	0.795	0.885					
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	E _t	48.240	-0.042	0.052	0.060	0.154	0.104					
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	dEr	48.240	-0.054	0.006	0.007	0.065	0.066					
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	dE_{t-1}	48.240	-0.055	0.010	0.008	0.076	0.070					
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	dNA.	48 240	-0.120	0.062	0.052	0.280	0.185					
RD_t $R2_{t+1}$ $R2_{t+2}$ RD_t	dNA.	48 240	-0.123	0.092	0.045	0 341	0 2 5 5					
dk_{t} dk_{240} -0.001 0.001 0.000 0.005 0.012 dRD_{t+1} 48.240 -0.001 0.001 0.000 0.006 0.012 l_{t} 48.240 -0.001 0.001 0.000 0.006 0.012 dl_{t} 48.240 -0.008 0.001 0.000 0.010 0.010 dl_{t+1} 48.240 -0.008 0.001 0.000 0.011 0.011 D_{t} 48.240 -0.008 0.001 0.000 0.011 0.011 D_{t} 48.240 -0.008 0.001 0.000 0.016 0.023 dD_{t+1} 48.240 -0.010 0.002 0.000 0.016 0.023 C_{t} 48.240 -0.011 0.003 0.000 0.016 0.023 C_{t} 48.240 -0.016 0.012 0.002 0.091 0.023 C_{t+1} 48.240 -0.016 0.012 0.002 0.082 0.082 dC_{t+1} 48.240 0.009 0.126 0.073 0.314 0.148 dC_{t} 48.247 -0.065 0.005 0.002 0.091 0.094 Panel C: Firm-level variables used to measure the target cash levelIn Real/NA _t 28.477 -0.078 0.015 0.034 0.118 0.195 $VOLA_t$ 28.477 -0.078 0.061 0.057 0.306 0.195 0.073 0.306 0.195	RD.	48 240	0.000	0.016	0.000	0.053	0.042					
dRD_{t+1} $48,240$ -0.001 0.001 0.000 0.006 0.012 l_t $48,240$ -0.002 0.020 0.016 0.043 0.019 dl_t $48,240$ -0.008 0.001 0.000 0.010 0.010 dl_{t+1} $48,240$ -0.008 0.001 0.000 0.011 0.011 D_t $48,240$ -0.008 0.001 0.000 0.016 0.022 dD_t $48,240$ -0.010 0.002 0.000 0.016 0.020 dD_{t+1} $48,240$ -0.011 0.003 0.000 0.019 0.023 C_t $48,240$ -0.011 0.003 0.000 0.019 0.023 dC_t $48,240$ -0.065 0.005 0.002 0.082 0.082 dC_t $48,240$ -0.065 0.012 0.002 0.091 0.94 Panel C: Firm-level variables used to measure the target cash level $IIRealNA_t$ $28,477$ -0.078 0.015 0.034 0.118 0.149 NWC_t $28,477$ -0.156 0.061 0.057 0.306 0.195 0.002 0.082 0.1149 NWC_t $28,477$ 0.055 0.125 0.106 0.221 0.073 $R)/SALES_t$ $28,477$ 0.055 0.125 0.106 0.221 0.073 $RO/SALES_t$ $28,477$ 0.064 2.338 13.100 2.780 1.070 3.4100 $CAREX_t$ $28,477$ 0.0	dRD.	48 240	_0.000	0.010	0.000	0.005	0.012					
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	dPD.	48 240	-0.001	0.001	0.000	0.005	0.012					
i_t $i_{0,2+0}$ 0.002 0.001 0.010 0.010 dl_t $48,240$ -0.008 0.001 0.000 0.011 0.010 dl_{t+1} $48,240$ -0.008 0.001 0.000 0.011 0.011 D_t $48,240$ -0.000 0.018 0.008 0.048 0.020 dD_t $48,240$ -0.010 0.002 0.000 0.016 0.020 dD_{t+1} $48,240$ -0.011 0.003 0.000 0.019 0.023 C_t $48,240$ -0.065 0.005 0.002 0.082 0.082 dc_t $48,240$ -0.065 0.005 0.002 0.091 0.094 dc_t $48,240$ -0.065 0.012 0.002 0.091 0.094 dc_t $48,241$ -0.065 0.012 0.002 0.091 0.094 $Panel C: Firm-level variables used to measure the target cash level10.0020.0010.094Panel C: Firm-level variables used to measure the target cash level10.0150.0130.0160.221NWC_t28,4770.0780.0150.0340.1180.149NWC_t28,4770.0550.1250.1060.2210.073NV_t28,4770.0642.33813.1002.7801.070SALES_t28,4770.0160.0740.0550.1540.068L$		48 240	0.002	0.001	0.000	0.000	0.012					
dl_{t+1} 48,240-0.0080.0010.0000.0110.011 D_t 48,240-0.0100.0020.0000.0160.020 dD_t 48,240-0.0110.0020.0000.0160.020 dL_{t+1} 48,240-0.0110.0030.0000.0190.023 C_t 48,2400.0090.1260.0730.3140.148 dc_t 47,967-0.0650.0050.0020.0910.094Panel C: Firm-level variables used to measure the target cash levelIn RealNAt28,47710.70013.00012.90015.5001.800FCF_t28,477-0.0780.0150.0340.1180.195VOLAt28,4770.0550.1250.1060.2210.073RD/SALESt28,4770.0550.1250.1060.2210.073RD/SALESt28,4770.0000.0300.0000.0820.114MVt28,4770.0550.1250.1060.2210.073RD/SALESt28,4770.0160.0740.0550.1540.068LEVERAGEt28,4770.0160.0740.0550.1540.068LEVERAGEt28,4770.0160.0740.0550.1540.068LEVERAGEt28,4770.0160.0740.0550.1540.068LEVERAGEt28,4770.0000.7021.0001.0000.457	dI.	48 240	-0.002	0.020	0.010	0.045	0.015					
u_{t+1} $10,240$ $-0,000$ 0.001 0.000 0.011 0.001 D_t $48,240$ -0.010 0.002 0.000 0.016 0.020 dD_t $48,240$ -0.011 0.003 0.000 0.019 0.023 C_t $48,240$ -0.011 0.003 0.000 0.019 0.023 C_t $48,240$ 0.009 0.126 0.073 0.314 0.148 dC_t $47,967$ -0.065 0.005 0.002 0.082 0.082 dC_{t+1} $48,041$ -0.065 0.012 0.002 0.091 0.994 Panel C: Firm-level variables used to measure the target cash levelIn RealNA _t $28,477$ 10.700 13.000 12.900 15.500 1.800 FCF_t $28,477$ -0.078 0.015 0.034 0.118 0.149 NWC _t $28,477$ -0.055 0.125 0.106 0.221 0.073 $VOLA_t$ $28,477$ 0.000 0.030 0.000 0.082 0.114 MV_t $28,477$ 0.065 0.125 0.106 0.221 0.073 $SALESc_t$ $28,477$ 0.016 0.074 0.055 0.154 0.068 $LEVERAGE_t$ $28,477$ 0.016 0.074 0.055 0.154 0.068 $LEVERAGE_t$ $28,477$ 0.016 0.702 1.000 1.000 0.477 $DIVDUMMMY_t$ $28,477$ 0.000 0.702 1.00	di	40,240	-0.008	0.001	0.000	0.010	0.010					
D_t 46,2400.0000.0160.0060.0460.027 dD_t 48,240-0.0110.0020.0000.0160.020 dD_{t+1} 48,240-0.0110.0030.0000.0190.023 C_t 48,2400.0090.1260.0730.3140.148 dC_t 47,967-0.0650.0050.0020.0820.082 dC_{t+1} 48,041-0.0650.0120.0020.0910.094Panel C: Firm-level variables used to measure the target cash levelIn RealNA _t 28,47710.70013.00012.90015.5001.800FCF _t 28,477-0.0780.0150.0340.1180.149NWC _t 28,4770.0550.1250.1060.2210.073VOLA _t 28,4770.0550.1250.1060.2210.073SALES _t 28,4770.8642.33813.1002.7801.070SALESG _t 28,4770.0160.0740.0550.1540.068LEVERAGE _t 28,4770.0160.0740.0550.1540.068LEVERAGE _t 28,4770.0160.0740.0550.1540.068LEVERAGE _t 28,4770.0160.0740.0550.1540.068LEVERAGE _t 28,4770.0160.0740.0550.1540.068LEVERAGE _t 28,4770.0000.7021.0000.457	u_{t+1}	40,240	-0.000	0.001	0.000	0.011	0.011					
dD_t $43,240$ -0.010 0.002 0.000 0.016 0.020 dD_{t+1} $48,240$ -0.011 0.003 0.000 0.019 0.023 C_t $48,240$ 0.009 0.126 0.073 0.314 0.148 dC_t $47,967$ -0.065 0.005 0.002 0.082 0.082 dC_{t+1} $48,041$ -0.065 0.012 0.002 0.091 0.094 Panel C: Firm-level variables used to measure the target cash levelIn RealNA _t $28,477$ 10.700 13.000 12.900 15.500 1.800 FCF _t $28,477$ -0.078 0.015 0.034 0.118 0.149 NWC _t $28,477$ -0.055 0.125 0.106 0.221 0.073 RD/SALES _t $28,477$ 0.055 0.125 0.106 0.221 0.073 RD/SALES _t $28,477$ 0.864 2.338 13.100 2.780 1.070 SALESC _t $28,477$ 0.016 0.074 0.055 0.154 0.668 LEVERAGE _t $28,477$ 0.014 0.249 0.237 0.477 0.179 DIVDUMMY _t $28,477$ 0.000 0.702 1.000 1.000 0.457	dD	40,240	0.000	0.018	0.008	0.048	0.027					
ab_{t+1} $4b_{2}240$ -0.011 0.003 0.000 0.013 0.013 0.023 C_t $4b_{2}240$ 0.009 0.126 0.073 0.314 0.148 dC_t 47.967 -0.065 0.005 0.002 0.082 0.082 dC_{t+1} $4b_{0}041$ -0.065 0.012 0.002 0.091 0.094 Panel C: Firm-level variables used to measure the target cash levelIn RealNA_t $2b_{4}77$ 10.700 13.000 12.900 15.500 1.800 FC_T $2b_{4}477$ -0.078 0.015 0.034 0.118 0.149 NWC_t $2b_{4}477$ -0.0755 0.125 0.106 0.221 0.073 $RD/SALES_t$ $2b_{4}477$ 0.000 0.030 0.000 0.082 0.114 MV_t $2b_{4}477$ 0.864 2.338 13.100 2.780 1.070 $SALESC_t$ $2b_{4}477$ 0.016 0.074 0.055 0.154 0.668 $LEVERAGE_t$ $2b_{4}477$ 0.014 0.249 0.237 0.457		40,240	-0.010	0.002	0.000	0.010	0.020					
C_t 48,2400.0090.1200.0730.5140.148 dC_t 47,967-0.0650.0050.0020.0820.082 dC_{t+1} 48,041-0.0650.0120.0020.0910.094Panel C: Firm-level variables used to measure the target cash level $\ln RealNA_t$ 28,47710.70013.00012.90015.5001.800 FC_t 28,477-0.0780.0150.0340.1180.149 NWC_t 28,477-0.0550.1250.1060.2210.073 $RD/SALES_t$ 28,4770.0550.1250.1060.2210.073 $RD/SALES_t$ 28,4770.8642.33813.1002.7801.070 $SALESG_t$ 28,477-7.51040.0729.23046.30034.100 $CAPEX_t$ 28,4770.0160.0740.0550.1540.668 $LEVERAGE_t$ 28,4770.0140.2490.2370.4770.179 $DIVDUMMY_r$ 28,4770.0000.7021.0001.0000.457	dD_{t+1}	48,240	-0.011	0.005	0.000	0.019	0.023					
dC_t 47,507 -0.003 0.003 0.002 0.002 0.002 0.002 dC_{t+1} 48,041 -0.065 0.012 0.002 0.002 0.091 0.094 Panel C: Firm-level variables used to measure the target cash level $\ln RealNA_t$ $28,477$ 10.700 13.000 12.900 15.500 1.800 FCF_t $28,477$ -0.078 0.015 0.034 0.118 0.149 NWC_t $28,477$ -0.055 0.125 0.106 0.221 0.073 $VOLA_t$ $28,477$ 0.055 0.125 0.106 0.221 0.073 $VOLA_t$ $28,477$ 0.000 0.030 0.000 0.082 0.114 MV_t $28,477$ 0.064 2.338 13.100 2.780 1.070 $SALES_t$ $28,477$ 0.016 0.074 0.055 0.154 0.068 $LEVERAG_t$ $28,477$ 0.014 0.249 0.237 0.477 0.179 $DIVDUMMY_r$ $28,477$ 0.000 0.702 1.000 1.000 0.457		48,240	0.009	0.120	0.073	0.014	0.148					
uc_{t+1} $46,041$ -0.063 0.012 0.002 0.091 0.091 Panel C: Firm-level variables used to measure the target cash level $\ln RealNA_t$ $28,477$ 10.700 13.000 12.900 15.500 1.800 FCF_t $28,477$ -0.078 0.015 0.034 0.118 0.149 NWC_t $28,477$ -0.055 0.125 0.061 0.057 0.306 0.195 $VOLA_t$ $28,477$ 0.055 0.125 0.106 0.221 0.073 $RD/SALES_t$ $28,477$ 0.000 0.030 0.000 0.082 0.114 MV_t $28,477$ 0.864 2.338 13.100 2.780 1.070 $SALESG_t$ $28,477$ -7.510 40.072 9.230 46.300 34.100 $CAPEX_t$ $28,477$ 0.014 0.249 0.237 0.477 0.179 $LEVERAGE_t$ $28,477$ 0.000 0.702 1.000 1.000 0.457		47,907	-0.005	0.005	0.002	0.082	0.082					
Panel C: Firm-level variables used to measure the target cash level $\ln RealNA_t$ 28,47710.70013.00012.90015.5001.800FCF_t28,477-0.0780.0150.0340.1180.149NWC_t28,477-0.1560.0610.0570.3060.195VOLA_t28,4770.0550.1250.1060.2210.073RD/SALES_t28,4770.8642.33813.1002.7801.070SALESG_t28,477-7.51040.0729.23046.30034.100CAPEX_t28,4770.0140.2490.2370.4770.179DIVDUMMY_r28,4770.0000.7021.0001.0000.457	uc_{t+1}	46,041	-0.065	0.012	0.002	0.091	0.094					
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	Panel C: Firm-level variables used to	measure the target cash	level									
FCF_t28,477 -0.078 0.015 0.034 0.118 0.149 NWC_t28,477 -0.156 0.061 0.057 0.306 0.195 VOLA_t28,477 0.055 0.125 0.106 0.221 0.073 RD/SALES_t28,477 0.000 0.030 0.000 0.082 0.114 MV_t28,477 0.000 0.030 0.000 0.082 0.114 MV_t28,477 -7.510 40.072 9.230 46.300 34.100 CAPEX_t28,477 0.016 0.074 0.055 0.154 0.068 LEVERAGE_t28,477 0.014 0.249 0.237 0.477 0.179 DI/DUMMY_r28,477 0.000 0.702 1.000 1.000 0.457	ln RealNAt	28,477	10.700	13.000	12.900	15.500	1.800					
NWC t28,477 -0.156 0.0610.0570.3060.195VOLA 28,4770.0550.1250.1060.2210.073RD/SALES 28,4770.0000.0300.0000.0820.114MV 28,4770.8642.33813.1002.7801.070SALESG t28,477-7.51040.0729.23046.30034.100CAPEX LEVERAGE t28,4770.0160.0740.0550.1540.068LEVERAGE LEVERAGE LVDUMMY r28,4770.0000.7021.0001.0000.457	FCF _t	28,477	-0.078	0.015	0.034	0.118	0.149					
$VOLA_t$ 28,4770.0550.1250.1060.2210.073 $RD/SALES_t$ 28,4770.0000.0300.0000.0820.114 MV_t 28,4770.8642.33813.1002.7801.070 $SALESG_t$ 28,477-7.51040.0729.23046.30034.100 $CAPEX_t$ 28,4770.0160.0740.0550.1540.068 $LEVERAGE_t$ 28,4770.0140.2490.2370.4770.179 $DI/DUMMY_r$ 28,4770.0000.7021.0001.0000.457	NWCt	28,477	-0.156	0.061	0.057	0.306	0.195					
$RD/SALES_t$ 28,4770.0000.0300.0000.0820.114 MV_t 28,4770.8642.33813.1002.7801.070 $SALESG_t$ 28,477-7.51040.0729.23046.30034.100 $CAPEX_t$ 28,4770.0160.0740.0550.1540.068 $LEVERAGE_t$ 28,4770.0140.2490.2370.4770.179 $DI/DUMMY_r$ 28,4770.0000.7021.0001.0000.457	VOLAt	28,477	0.055	0.125	0.106	0.221	0.073					
MV_t 28,4770.8642.33813.1002.7801.070 $SALESG_t$ 28,477 -7.510 40.0729.23046.30034.100 $CAPEX_t$ 28,4770.0160.0740.0550.1540.068 $LEVERAGE_t$ 28,4770.0140.2490.2370.4770.179 $DIVDUMMY_t$ 28,4770.0000.7021.0001.0000.457	RD/SALES _t	28,477	0.000	0.030	0.000	0.082	0.114					
SALESGt28,477 -7.510 40.0729.23046.30034.100CAPEXt28,4770.0160.0740.0550.1540.068LEVERAGEt28,4770.0140.2490.2370.4770.179DIVDUMMYt28,4770.0000.7021.0001.0000.457	MV	28,477	0.864	2.338	13.100	2.780	1.070					
CAPEX _t 28,477 0.016 0.074 0.055 0.154 0.068 LEVERAGE _t 28,477 0.014 0.249 0.237 0.477 0.179 DIVDUMMY _t 28,477 0.000 0.702 1.000 0.457	SALESGt	28,477	-7.510	40.072	9.230	46.300	34.100					
LEVERAGE 28,477 0.014 0.249 0.237 0.477 0.179 DIVDUMMY, 28,477 0.000 0.702 1.000 1.000 0.457	CAPEXt	28,477	0.016	0.074	0.055	0.154	0.068					
DIVDUMMY, 28,477 0.000 0.702 1.000 0.457	LEVERAGE	28,477	0.014	0.249	0.237	0.477	0.179					
	DIVDUMMYt	28,477	0.000	0.702	1.000	1.000	0.457					

This table shows descriptive statistics of the model variables: number of observations (N), 10% and 90% percentiles, mean, median, and standard deviation (S.D.). Panel A describes the main proxy for information asymmetry, denoted as DISPM, on a country-level. DISPM measures the scaled dispersion (standard deviation) of 1-year analysts' earnings per share forecasts provided by I/B/E/S. A higher value of DISPM indicates more pronounced information asymmetry. As an alternative measure of information asymmetry, ERRORF denotes the difference between actual and forecasted earnings per share, scaled by the median earnings per share forecast. IA-INDEX is a comprehensive index of information asymmetry based on the various dimensions of the concept. In addition to ERRORF, this index is based on quintile rankings of firm size, R&D expenditure, Tobin's Q, and the number of analysts following the firm in a given year. The index values range from 5 (lowest information asymmetry) to 25 (highest information asymmetry). Panel B shows the variables that enter the main valuation regression in Eq. (4). The data set covers the period from 1995 to 2005. All firms from the different countries are included for which I/B/E/S provides analysts' forecasts and for which company data is available from Worldscope (except financial firms and utilities). The main proxy for information asymmetry, DISPM, is based on the standard deviation of analysts' earnings per share forecasts (analysts' forecasts dispersion), and this measure can only be computed when the forecasts are at least based on two analysts. A firm is omitted from the sample if DISPM cannot be calculated in at least one sample year, i.e., if this firm is not covered by at least two analysts in at least one sample year. All variables are trimmed at the 1% and the 99% tails. Vt denotes the total market value of the firm (market value of equity plus book value of debt); E_t is earnings before interest and extraordinary items (after depreciation and taxes); NA_t is net assets (book value of total assets minus cash); RD_t is research and development (R&D) expenditure; I_t is interest expense; D_t is total dividends paid; and C_t is cash holdings in year t. $dX_t = X_t - X_{t-1}$ denotes the past 1-year change of variable X_t , and $dX_{t+1} = X_{t+1} - X_t$ is the future 1-year change of variable X_t . All variables are scaled by total assets (A_t). Panel C shows the variables that are used to compute the target cash level (and hence excess cash) in the alternative specification of the valuation regression in Eq. (6). lnRealNA_t is the natural logarithm of net assets (book value of total assets minus cash) in US dollar terms for the year 2000; FCF_t is operating income after interest and taxes; NWC_t is working capital minus cash; VOLA_t is the standard deviation of a firm's monthly stock returns over the prior 12 months; MV_t is the market value of the firm, computed as the number of shares outstanding times share price plus total liabilities; SALESG_t is sales growth; CAPEX_t is capital expenditure; and LEVERAGE_t is total debt (interest bearing) divided by total assets. All variables are scaled as shown in the cash target model in Eq. (5). DIVDUM_t denotes a dividend dummy variable, which is set equal to one if the firm paid dividends or engaged in share repurchases, and zero otherwise.

a firm. Greater disagreement among analysts indicates a higher level of information asymmetry. Supporting this notion, Diether et al. (2002) observe a negative relationship between the dispersion of analysts' forecasts and subsequent stock returns. They argue that the dispersion of analysts' forecasts is not merely a proxy for risk but rather a metric for differences of opinion. D'Mello and Ferris (2000) report stronger announcement effects for firms whose forecasts exhibit higher dispersion, also suggesting that analysts' dispersion captures information asymmetry. Finally, Autore and Kovacs (2010) document that firms avoid to raise external funds in states with a high dispersion of analysts' forecasts.

To compute the dispersion of analysts' forecasts, we use 1-year consensus forecasts of the earnings per share provided by I/B/E/S. The dispersion of these forecasts (defined as the firm-level standard deviation of all forecasts across covering analysts) is not updated in each month for every firm. If we took the data only for one specific month, we would lose all firm-year observations for which no (updated) estimate for this particular month is available. Therefore, we calculate the average of the monthly dispersions in each year. In order to make this measure comparable across firms, the standard deviation of forecasts needs to be scaled. As our dependent variable (the firm value) is related to the stock price. we scale by the median forecast rather than the stock price in order to avoid an endogeneity problem. By adding one and taking the natural logarithm, our measure converges to a normal distribution. Therefore, our main proxy for information asymmetry, denoted as DISPM, is:

$$DISPM = \ln\left(1 + \frac{Standard \ deviation \ of \ analysts' \ forecasts}{|Median \ forecast|}\right), \quad (1)$$

where the standard deviation is the mean of the monthly standard deviations taken over the entire year. A more detailed version of this formula is presented in the Appendix. A caveat with this variable is that there could be systematic differences in the level of dispersion across countries. Harford et al. (2008) argue that countryeffects (e.g., their legal settings and transparency standards) in international studies may dominate firm-level limitations on shareholder rights. For example, in our setup high asymmetric information levels in the United States or the United Kingdom could be similar to low levels of information asymmetry in countries with poor shareholder rights or high corruption. Panel A of Table 2 presents descriptive statistics of the variable DISPM for different countries. The most important observation is that mean dispersion is higher in emerging stock markets compared to developed stock markets. However, there are even large differences across G-7 countries, with mean dispersion being very high in Germany and Italy and the lowest in the United States and the United Kingdom. These findings seem consistent with Capstaff et al. (2001), who argue that differences in the performance of analysts' forecasts are due to differences in earnings behavior, accounting practices, and influences of securities markets. In their European sample, analysts' forecasts are most accurate in the United Kingdom and least accurate in Italy. The high mean dispersion in Germany could be consistent with the findings in Bessler and Stanzel (2009). They document that equity research in a universal banking system suffers from particularly pronounced conflicts of interest. More generally, Harford et al. (2008) argue that country-level effects dominate firm-level effects in international corporate governance studies. Therefore, to account for potential biases that may result from country-level effects in our analysis, we construct a dummy variable which takes the value of one (high information asymmetry) if a firm exhibits a value of DIS-PM above its country median in a given year or if the firm is not covered by at least two analysts, and hence we cannot compute DISPM, and zero (low information asymmetry) otherwise. This variable is denoted as DISPM-DUMMY. When there are few or no analysts following a firm, then little information is generated, and hence the level of information asymmetry is high (Chang et al., 2006). Therefore, an appealing property of the DISPM-DUMMY variable is that it exploits the full sample of 48,240 firm-year observations rather than the 34,876 firm-year observations for which we are able to compute our dispersion measure.

As another robustness test, we follow Cai et al. (2009) and use a comprehensive measure by constructing an index of information asymmetry based on the various dimensions of the concept. For example, Krishnaswami and Subramaniam (1999) suggest the error in analysts' forecasts, defined as the difference between the mean forecast and the actual earnings per share, as a measure of information asymmetry. In addition, the empirical literature has introduced various simpler measures of information asymmetry. For example, information asymmetry tends to decrease with firm size (Vermaelen, 1981), increase with R&D expenditure (Aboody and Lev, 2000), increase with growth opportunity (Smith and Watts, 1992) and decrease with analyst coverage (Krishnaswami and Subramaniam, 1999). Although these measures might be partially correlated, each contains unique information. Specifically, our information asymmetry index is based on the rankings of the error in analysts' forecasts, firm size, R&D expenditure, Tobin's O. and the number of analysts following a firm:

Error in analysts' forecasts: Elton et al. (1984) provide evidence that most of the forecast error in the last month of the fiscal year can be explained by misestimation of firm-specific factors rather than by misestimation of economy-wide or industry factors. We use the following measure of the error in analysts' forecasts:

$$ERRORF = \ln\left(1 + \frac{|EPS_{Forecast} - EPS_{Actual}|}{|Median \ EPS|}\right),\tag{2}$$

where the forecast of earnings per share, labeled *EPS*_{Forecast}, is the average of all forecasts for a firm provided by analysts in November and December of the previous year. The difference between actual and forecasted earnings per share is scaled by the median earnings per share forecast.

Firm size: Large firms may face less information asymmetry because they tend to be more mature, have established disclosure policies, and receive more attention from the market (Diamond and Verrecchia, 1991; Ozkan and Ozkan, 2004). We use total assets to measure firm size.

R&D expenditure: Analyzing insider trading gains in firms with high and low R&D expenses, Aboody and Lev (2000) argue that R&D is related to information asymmetry. In order to measure a firm's R&D intensity, we use a dummy variable that takes the value of one if the firm reports R&D expenses, and zero if they are zero or missing.

Tobin's Q: Information asymmetry is more severe for firms with significant growth opportunities (Smith and Watts, 1992). Therefore, proxies for firms' investment opportunities have been used to measure information asymmetry (McLaughlin et al., 1998). We use Tobin's Q to measure growth opportunities, computed as book value of assets minus book value of equity plus market value of equity divided by book value of assets. *Number of analysts following the firm:* The number of analysts is another proxy for the supply of information about a firm. Presumably, the more analysts follow a firm, the more information is discovered and revealed to the public, and hence asymmetric information is limited (Chang et al., 2006). Brennan and Subrahmanyam (1995) argue that greater analyst coverage tends to reduce the adverse selection costs as measured by the inverse of market depth.

In order to construct the information asymmetry index, denoted as *IA-INDEX*, we first calculate a firm's quintile ranking over all firms for each dimension of information asymmetry in a given year. A higher score indicates a greater degree of information asymmetry. For example, a firm receives a score of 5 (1) if it belongs to the 20% smallest (largest) firms in a given year. We then add up the ranks along all five dimensions of information asymmetry.⁷ Therefore, the largest (smallest) value the variable *IA-INDEX* can take is 25 (5) for the firms with the highest (lowest) degree of information asymmetry. Panel A of Table 2 shows descriptive statistics of this information asymmetry index.

3.1.3. Sample splits

We divide our sample into several subgroups to test whether financial constraints and corporate governance structures have an impact on how information asymmetry influences the value of cash. This approach allows us to differentiate between Hypothesis 1 (related to costly external finance due to adverse selection) and Hypothesis 2 (related to the free cash flow problem).

In order to investigate the influence of financial constraints, we use two measures at the country-level (stock market capitalization to GDP and private bond market capitalization to GDP) and one variable at the firm-level (firm size) to split the sample. Specifically, stock market capitalization to GDP is the ratio of the market value of listed shares in a country to its gross domestic product. Private bond market capitalization to GDP is the ratio of a country's private domestic debt securities (issued by financial institutions and corporations) to its gross domestic product. Countries with higher ratios tend to have more developed capital markets, where firms have better access to capital and are less constrained. Both ratios of the stock market capitalization to GDP and the private bond market capitalization to GDP are taken from the website of Ross Levine.⁸ According to Almeida et al. (2004), small firms also tend to be constrained. Therefore, in addition to the country-level measures, we use firm size as a firm-level variable to split the sample; it is measured by the firm's equity market capitalization.

In order to investigate the influence of corporate governance, we use four measures at the country-level (anti-director rights index, rule of law index, corruption index, and legal system classification) and one variable at the firm-level (percentage of closely held shares) to split the sample. The three country-level indices are related to the two components of investor protection. The first is a legal component, where investors are granted legal rights, and the second is an enforcement component, where the quality of a country's institutions determines the extent to which these rights are respected and enforced. The anti-director rights index is a measure of shareholder protection. It consists of six components, of which three are concerned with shareholder voting (voting by mail, voting without blocking of shares, and calling an extraordinary meeting), and three with minority protection (proportional board representation, preemptive rights to new issues, and judicial remedies). Depending on the fulfillment of these criteria, the index ranges from zero to six, where higher index values indicate better protection of minority shareholder rights. The anti-director rights index is provided on the website of Rafael La Porta.⁹ The rule of law index and the corruption index both measure the quality of institutions that support the rights of investors. The rule of law index captures the extent to which agents have confidence in the rules of society, the quality of contract enforcement, and the courts. The corruption index measures the extent to which public power is used to extract private gains. Index levels range from negative to positive, where firms in countries with lower or even negative index values generally operate under weaker corporate governance structures, and it is more difficult for investors to make use of their formal rights. The rule-of-law index and the corruption index are constructed by the Worldbank. In addition, La Porta et al. (1998) document that in common law countries minority shareholders are better protected against expropriation by insiders compared to civil law countries. The legal system classification is again provided on the website of Rafael La Porta. Finally, to mitigate the problems associated with managerial opportunism, Jensen and Meckling (1976) suggest that firms increase managerial equity ownership. Therefore, in addition to the country-level measures, we use the percentage of closely held shares as a firm-level variable to split the sample. Following Morck et al. (1988) and Opler et al. (1999), we choose three cut-off levels for ownership of corporate insiders: 0–5%, 5–25%, and 25% or more.

In our main model specification, we use median-splits (where applicable) of our sample based on data for the year 2000, which is the year in the middle of our sample period. In robustness tests, we also use different years at the beginning and the end of the sample period to split the sample. A limitation of our analysis is that there is not always a clear-cut distinction between the variables that are used to split the sample according to the corporate governance structure and those that are used to classify firms as financially constrained. The legal system is used as a proxy for the quality of corporate governance. Civil law countries generally have narrower capital markets than common law countries (La Porta et al., 1998), and hence the legal system could also be associated with the degree of financial constraints.

3.2. Empirical methodology

While our study is, to the best of our knowledge, the first that investigates the influence of information asymmetry on the value of cash, it is not the first one that analyzes the value of cash in different settings. Fama and French (1998) study the impact of debt and dividends on firm value. Pinkowitz et al. (2006) and Dittmar and Mahrt-Smith (2007) modify their valuation model to estimate the marginal value of cash. Both studies put the emphasis on the relationship between cash and corporate governance. In our main regression specification, we extend the model of Pinkowitz et al. (2006) to test the impact of information asymmetry on the value of cash. In order to check the robustness of our results, we also build on the approach of Dittmar and Mahrt-Smith (2007).¹⁰ This section describes these two methods and shows how we modify them to estimate the relationship between the market value of cash and firm-specific and time-varying information asymmetry.

3.2.1. Main valuation regression

The starting point is the valuation model in Fama and French (1998), who examine the influence of debt and dividends on firm value. To estimate the value of cash, Pinkowitz et al. (2006) adapt their framework. Most important, they split up the changes in assets into its cash and non-cash components. To estimate the relationship between market value and cash holdings, they use the following valuation model:

$$V_{t} = \alpha + \beta_{1}E_{t} + \beta_{2}dE_{t} + \beta_{3}dE_{t+1} + \beta_{4}dNA_{t} + \beta_{5}dNA_{t+1} + \beta_{6}RD_{t} + \beta_{7}dRD_{t} + \beta_{8}dRD_{t+1} + \beta_{9}I_{t} + \beta_{10}dI_{t} + \beta_{11}dI_{t+1} + \beta_{12}D_{t} + \beta_{13}dD_{t} + \beta_{14}dD_{t+1} + \beta_{15}dV_{t+1} + \beta_{16}C_{t} + \varepsilon_{t},$$
(3)

where V_t denotes the total market value of the firm (market value of equity plus book value of debt); E_t is earnings before interest and extraordinary items (after depreciation and taxes); NA_t is net assets

⁷ This procedure follows Peyer and Vermaelen (2009), who construct an undervaluation index to examine the motives for share repurchases. They argue that the rule of equally weighting the characteristics is arbitrary. However, the idea is to test whether the correlation between the factors leads to a significant improvement in identifying firms with high information asymmetry by taking into account some potential for cross-correlation.

⁸ See http://www.econ.brown.edu/fac/Ross_Levine/Publications.html.

⁹ See http://www.mba.tuck.dartmouth.edu/pages/faculty/rafael.laporta/publications.html and La Porta et al. (1998).

¹⁰ Another approach to estimate the value of cash (not used in this study) is the method of Faulkeneder and Wang (2006). They regress the cash ratio (in levels and differences) on excess stock returns.

(book value of total assets minus cash); RD_t is research and development (R&D) expenditure; I_t is interest expense; D_t is total dividends paid; and C_t is cash holdings at time t. $dX_t = X_t - X_{t-1}$ denotes the past 1-year change of variable X_t , and $dX_{t+1} = X_{t+1} - X_t$ is the future 1-year change of variable X_t . All variables are scaled by total assets (A_t). The dependent variable is the spread of value over cost. The control variables (in levels and differences) are included into the model to capture expectations about future earnings and other effects that potentially influence the value of the firm.¹¹ The most important coefficient is β_{16} on the level of cash, which measures the sensitivity of firm value to a one-dollar increase in cash holdings. Assuming that the impact of a change in cash on future cash flows is measured by the lead variables that capture investors' expectations, the coefficient on cash holdings is an estimate of the market value of a marginal dollar of cash.

As specified in our hypotheses, we are ultimately interested in the value of cash in connection with firm-specific and time-varying information asymmetry. In order to measure this dynamic effect, an additional interaction term is included into the valuation model. This variable is calculated by multiplying the cash level ($C_{i,t}$) by the dispersion of analysts' earnings forecasts (*DISPM*_{i,t}). Dispersion itself is also used as an explanatory variable to control for a direct influence of information asymmetry on firm value. We use a fixed effects estimator to focus on the within-dimension of the data. To control for macroeconomic effects, we also include time dummy variables into the model. This results in our main testable model:

$$V_{i,t} = \alpha + \beta_{1}E_{i,t} + \beta_{2}dE_{i,t} + \beta_{3}dE_{i,t+1} + \beta_{4}dNA_{i,t} + \beta_{5}dNA_{i,t+1} + \beta_{6}RD_{i,t} + \beta_{7}dRD_{i,t} + \beta_{8}dRD_{i,t+1} + \beta_{9}I_{i,t} + \beta_{10}dI_{i,t} + \beta_{11}dI_{i,t+1}\beta_{12}D_{i,t} + \beta_{13}dD_{i,t} + \beta_{14}dD_{i,t+1} + \beta_{15}dV_{i,t+1} + \beta_{16}C_{i,t} + \beta_{17}(C \times DISPM)_{i,t} + \beta_{18}DISPM_{i,t} + \alpha_{i} + \mu_{t} + \varepsilon_{i,t},$$
(4)

where α_i and μ_t denote entity- and time-fixed effects. The coefficient of interest in our main valuation regression is β_{17} . This coefficient on the interaction term measures the market value of a marginal dollar of cash in connection with firm-specific and time-varying information asymmetry. While a positive coefficient would support Hypothesis 1 (related to the pecking order theory), a negative one would support Hypothesis 2 (related to the free-cash flow theory).

Statistical inference of the model is based on Driscoll and Kraay (1998) standard errors. Höchle (2007) documents that these standard errors are robust to very general forms of cross-sectional and temporal dependence. Alternatively, we also estimate the model using the Fama–MacBeth approach. While this latter method is commonly used in the empirical corporate finance literature, Petersen (2009) forcefully shows that it cannot control for cross-sectional dependence. The reduced model in Eq. (3) exploits the full sample of 48,240 firm-year observations. As firm-year observations for which *DISPM* is not defined drop out of the sample, the number of firm-year observations decreases to 34,876 in the model in Eq. (4) that includes the interaction term. Panel B of Table 2 presents the descriptive statistics of all model variables. The values are similar to those in Pinkowitz et al. (2006).

3.2.2. An alternative specification: Incorporating the target cash level As a robustness test, we estimate an extended version of the model in Dittmar and Mahrt-Smith (2007). Instead of the actual cash level, they use the level of excess cash as the independent variable. Therefore, we follow Opler et al. (1999) and predict the normal level of cash that is needed for operations or investments. In order to control for the transaction costs motive in the estimation of the target cash level, we include net assets (total assets minus cash), net working capital, and a proxy for cash flow volatility. In addition, in order to account for the precautionary motive (Myers and Majluf, 1984), we use investment opportunities (market-to-book ratio), cash flow, and access to external capital as measured by firm size (book value of assets in US dollars terms for the year 2000). As suggested by Dittmar and Mahrt-Smith (2007), an endogeneity problem occurs if the raw market-to-book ratio is used to predict the normal or target level of cash in order to calculate excess cash, and excess cash is then taken to predict the market-to-book ratio. They instrument the market-to-book ratio with past sales growth (SALESG) and use this instrumented market-tobook ratio to predict cash holdings. We instrument the marketto-book ratio using the average of last year's and current year's sales growth. Following Opler et al. (1999), we also include capital expenditures, leverage, and a dividend dummy variable. The regression model to estimate the normal level of cash is:

$$\ln\left(\frac{C_t}{NA_t}\right) = \alpha + \beta_1 \ln(\text{RealNA}_t) + \beta_2 \frac{FCF_t}{NA_t} + \beta_3 \frac{NWC_t}{NA_t} + \beta_4 \text{VOLA}_t + \beta_5 \frac{\widehat{MV}_t}{TA_t} + \beta_6 \frac{RD_t}{SALES_t} + \beta_7 \frac{CAPEX_t}{NA_t} + \beta_8 \frac{DEBT_t}{TA_t} + \beta_9 DIVDUM_t + SECTDUM + \alpha_i + \varepsilon_t,$$
(5)

where *NA_t* is net assets (book value of total assets minus cash); $ln(RealNA_t)$ is the natural logarithm of net assets in US dollar terms for the year 2000; FCF_t is operating income after interest and taxes; NWC_t is working capital minus cash; $VOLA_t$ is the standard deviation of a firm's monthly stock returns over the prior 12 months; $\widehat{MV_t}$ is the market value of the firm, computed as the number of shares outstanding times share price plus total liabilities (instrumented with the average of last year's and current year's sales growth, $SALESG_t$); $CAPEX_t$ is capital expenditure; and $DEBT_t$ is total (interest bearing) debt. DIVDUM_t denotes a dividend dummy variable, which is set equal to one if a firm paid dividends or engaged in share repurchases, and zero in all other cases. SECTDUM denotes sector dummy variables.¹² The predictive regression is estimated using a fixed effects approach, where α_i denotes entity-fixed effects. Because firms from different countries will have different reasons to hold cash (e.g., due to legal differences that require cash buffers against adverse events), we follow Fresard and Salva (forthcoming) and estimate the target cash level independently for each country in our sample. The residual from these regressions, i.e., the difference between the actual and the exponential of the predicted log cash level, is defined as excess cash and labeled EXCASH. This variable is used to calculate the market value of excess cash in connection with firm-specific and time-varying information asymmetry:

$$V_{i,t} = \alpha + \beta_{1}E_{i,t} + \beta_{2}dE_{i,t} + \beta_{3}dE_{i,t+1} + \beta_{4}dNA_{i,t} + \beta_{5}dNA_{i,t+1} + \beta_{6}RD_{i,t} + \beta_{7}dRD_{i,t} + \beta_{8}dRD_{i,t+1} + \beta_{9}I_{i,t} + \beta_{10}dI_{i,t} + \beta_{11}dI_{i,t+1} + \beta_{12}D_{i,t} + \beta_{13}dD_{i,t} + \beta_{14}dD_{i,t+1} + \beta_{15}dV_{i,t+1} + \beta_{16}EXCASH_{i,t} + \beta_{17}(EXCASH \times DISPM)_{i,t} + \beta_{18}DISPM_{i,t} + \alpha_{i} + \mu_{t} + \varepsilon_{i,t},$$
(6)

where α_i and μ_t again denote entity- and time-fixed effects. Following Dittmar and Mahrt-Smith (2007), all variables are scaled by net assets, and the regression is estimated for positive values of excess cash using fixed effects and Driscoll and Kraay (1998) standard errors. Again, the main coefficient of interest is β_{17} . A positive

¹¹ In an alternative setting, Pinkowitz et al. (2006) use changes of cash instead of the cash level. However, they note that an increase in cash holdings could also lead to changes in expectations about future growth.

¹² There is some variance in the within dimension of the sample firms' sector classification, and hence *SECTDUM* does not drop out in a fixed effects regression. Our results are robust when we exclude the sector dummy variables.

coefficient on the interaction term supports Hypothesis 1 (related to the pecking order theory), and a negative one supports Hypothesis 2 (related to the free-cash flow theory).

We use the alternative model in Eq. (6) only as a robustness test of our main valuation regression in Eq. (4) for two reasons. First, Hypothesis 1 is based on the pecking order theory. In a strict pecking order world, there is no optimal level of cash. However, this alternative approach requires calculating excess cash as the deviation from a normal cash level. Excess cash encompasses the components of cash that cannot be directly related to operational needs or investment opportunities. It refers to the amount of cash holdings that can neither be justified based on the transaction cost motive nor on the precautionary motive. One would expect that it is held for discretionary purposes and is especially prone to managerial squandering. Therefore, excess cash may be strongly related to Jensen's (1986) free cash flow hypothesis, making it more difficult to disentangle our two opposing hypotheses. Second, the calculation of excess cash requires variables that are not available for all firms, thereby reducing our sample size. Due to data availability in the estimation of the normal cash level, our sample reduces to 28,477 firm-year observations (and even further in the interaction model). To save space, descriptive statistics of this reduced sample are omitted. The values are in line with those reported in Dittmar and Mahrt-Smith (2007). Panel C of Table 2 shows descriptive statistics of the variables used to predict target cash levels.

4. Empirical tests of the hypotheses

4.1. Main empirical results

Table 3 presents the results of our reduced valuation regression in Eq. (3) without information asymmetry. These results provide the basis for a comparison with previous studies that do not incorporate the influence of information asymmetry. The main coefficients of interest in Table 3 are those on cash (C_t) and on changes in cash (dC_t) . We focus on the fixed effects model that includes the level of cash as an explanatory variable. Using all firms in the sample, the estimated coefficient on cash in Panel A is 0.661; it is highly significant and can be interpreted as the market value of an additional dollar of cash. One reason why the market value of a dollar of cash is below one could be the impact of taxes on payouts (in particular, for firms with few investment opportunities). The comparable coefficient in Pinkowitz and Williamson (2004) is 1.05, but they only analyze a US sample. Pinkowitz et al. (2006) use international data, but they only report the corresponding coefficient for subgroups based on country characteristics rather than for their whole sample. Moreover, they only apply the Fama-MacBeth approach. Depending on the subgroup, they report coefficients that range between -0.03 and 1.24.13 Overall, our estimated coefficients fall into this range, although they are higher (and well above one) in the Fama-MacBeth approach.

Continuing with the results in Panel A, most other coefficients have the expected signs, and many are similar in magnitude to those in Pinkowitz and Williamson (2004) and Pinkowitz et al. (2006). Nevertheless, there are several differences. For example, Pinkowitz and Williamson (2004) present a positive coefficient on the earnings variable (E_t) in the Fama–MacBeth model compared to a negative one in the fixed effects specification. In contrast, we report positive coefficients in both specifications. Another observation is that the coefficient on earnings changes (dE_t) is negative in the fixed effects model and positive when the Fama– MacBeth approach is used. However, only the positive coefficient is statistically significant. An explanation is that the Fama–MacBeth approach cannot control for firm-fixed effects.

Panels B and C of Table 3 presents the results for subsamples of firms from developed stock markets and emerging stock markets. Presumably, firms from the latter subgroup suffer from poor investor protection and generally lower corporate governance standards. In line with expectations, the coefficients on C_t and dC_t are lower for firms in emerging markets than in developed markets. This supports the notion in Pinkowitz et al. (2006) that cash is worth less in countries with a low level of development because these countries suffer from poor investor protection.¹⁴

The main research question of our analysis is to assess whether the impact of firm-specific and time-varying information asymmetry on the market value of a dollar of cash is positive or negative. Panel A of Table 4 presents the results of our main valuation regression in Eq. (4) that includes the dispersion of analysts' earnings forecasts (DISPM), a measure of firm-specific and time-varving information asymmetry, and its interaction term with cash holdings. The most important observation is that the coefficient on the interaction term is significantly negative. We interpret this result as support for Hypothesis 2, suggesting that cash holdings have less value for a firm in states with a high degree of information asymmetry. The free cash flow problem seems to be more relevant for cash holdings in connection with information asymmetry than the advantage of having a liquidity reserve in states when the adverse selection costs of raising external funds are prohibitive. In order to examine whether this negative effect of information asymmetry on liquidity is also economically significant, we calculate the marginal value of cash conditional on the level of information asymmetry. Including an interaction term into the analysis, the market value of an additional dollar of cash is:

$$\frac{V}{A} = \alpha + \dots + \beta_{16} \frac{C}{A} + \beta_{17} \left(\frac{C}{A} \times DISPM \right) + \beta_{18} DISPM, \tag{7}$$

$$\frac{\partial \frac{V}{A}}{\partial \frac{C}{A}} = \frac{\partial V}{\partial C} = \beta_{16} + \beta_{17} DISPM.$$
(8)

Considering the results of the fixed effects model, the coefficient on cash is 0.786 and that on the interaction term is -0.465. Based on the median value of *DISPM* (0.112; Panel A of Table 2), the market value of an additional dollar of cash is $0.734 (= 0.786 - 0.465 \times 0.112)$. An increase in *DISPM* by one standard deviation (0.261; Panel A of Table 2) results in a marginal value of cash that is 0.121 dollar (or 16.5%) lower, and hence the market value of an additional dollar of cash decreases to 0.613. Accordingly, the negative effect of information asymmetry on the market value of cash is also economically significant.

Our results are confirmed when we use DISPM-DUMMY and IA-*INDEX* as alternative proxy variables for information asymmetry; the former accounts for biases that result from country-level effects of information asymmetry, and the latter serves as a comprehensive measure of information asymmetry along several dimensions of the concept. As shown in Panels B and C of Table 4, the coefficient on the interaction term is significantly negative in all specifications involving these two alternative measures. This result again supports Hypothesis 2, which posits that cash is less valuable in states with a high degree of information uncertainty. For example, the coefficient on the interaction term in the fixed effects model using DISPM-DUMMY indicates that being a high information asymmetry firm lowers the market value of the marginal dollar of cash by roughly 25 cents to 0.583 (= 0.827 - 0.244). In the spirit of Harford et al. (2008), this finding suggests that we do not simply capture country-level effects of granting and enforcing shareholder rights but also the effect of firm-level variation in

¹³ See Pinkowitz et al. (2006, Table V, p. 2742f).

¹⁴ We analyze the impact of investor protection in our sample splits in Table 5.

Estimated value of cash without information asymmetry.

	Panel A: All countries				Panel B: Developed markets				Panel C: Emerging markets			
	Fixed effec	ts	Fama-Mac	Beth	Fixed effect	ts	Fama-Mac	Beth	Fixed effects		Fama-MacBeth	
	Level	Diff.	Level	Diff.	Level	Diff.	Level	Diff.	Level	Diff.	Level	Diff.
Et	2.782***	2.596	1.460	1.050	2.916	2.708	1.524***	0.891	2.140	1.943	2.242***	2.081
	(30.07)	(18.20)	(9.45)	(6.36)	(29.95)	(18.57)	(9.53)	(4.82)	(11.65)	(10.53)	(4.67)	(4.33)
dEt	-0.070	-0.141	0.851	0.798	-0.041	-0.108	0.911	0.943	-0.111	-0.143	-0.109	-0.219
	(-1.17)	(-3.13)	(4.63)	(4.42)	(-0.53)	(-1.76)	(4.36)	(4.36)	(-1.44)	(-2.18)	(-0.39)	(-0.75)
dE_{t+1}	1.631	1.437	1.598	1.280	1.739	1.538	1.752	1.354	1.253	1.087	1.031	0.832
	(34.90)	(27.63)	(10.20)	(10.09)	(19.69)	(17.02)	(9.23)	(8.12)	(13.29)	(10.86)	(5.69)	(5.45)
dNAt	0.312	0.292	0.760	0.665	0.413	0.380	0.975	0.862	-0.018	-0.009	0.119	0.096
	(13.63)	(14.97)	(8.40)	(7.91)	(13.25)	(12.54)	(9.19)	(8.81)	(-1.02)	(-0.44)	(2.47)	(2.32)
dNA_{t+1}	0.573	0.636	0.467	0.555	0.595	0.675	0.483	0.599	0.483	0.514	0.357	0.397
	(8.61)	(10.28)	(4.91)	(5.40)	(7.61)	(9.55)	(4.77)	(5.41)	(11.38)	(12.12)	(4.60)	(5.30)
RD_t	3.590	3.391	5.295	7.005	3.706	3.497	4.773	6.683	-1.342	-1.105	10.508	10.927
	(15.02)	(10.20)	(14.93)	(14.26)	(14.98)	(10.54)	(14.18)	(13.72)	(-1.03)	(-0.81)	(4.97)	(5.36)
dRD_t	1.345	1.251	3.011	2.995	1.270	1.199	2.937**	3.005	2.824	2.709	0.573	0.088
	(3.95)	(4.78)	(3.35)	(3.05)	(3.57)	(4.23)	(3.22)	(2.90)	(4.14)	(3.82)	(0.25)	(0.04)
dRD_{t+1}	5.629	4.880	8.561	9.233	5.519	4.755	7.949***	8.837	3.128***	2.788***	11.568	11.256
	(14.01)	(26.53)	(9.18)	(11.76)	(13.53)	(20.61)	(8.46)	(11.05)	(3.39)	(2.67)	(6.00)	(6.43)
It	-0.351	-0.968***	1.159	-1.708**	-0.122	-1.631	3.586	-1.184^{*}	-0.703	-0.573	-0.744	-1.019
	(-1.46)	(-4.03)	(1.95)	(-2.54)	(-0.17)	(-2.29)	(7.78)	(-1.96)	(-0.98)	(-0.78)	(-0.76)	(-1.01)
dI_t	-0.964***	-0.876	-0.688	-0.034	-1.963	-1.681	-1.793	-1.040	-0.457	-0.474	0.023	0.280
	(-2.71)	(-2.79)	(-0.65)	(-0.05)	(-4.24)	(-3.18)	(-1.51)	(-0.84)	(-0.59)	(-0.61)	(0.03)	(0.39)
dI_{t+1}	-2.681	-3.445	-2.617**	-4.251	-3.988***	-5.350^{-10}	-3.234**	-6.108	-1.452	-1.628***	-1.076	-1.342
	(-10.01)	(-13.51)	(-3.07)	(-4.86)	(-4.52)	(-6.47)	(-2.64)	(-4.37)	(-3.69)	(-3.61)	(-1.24)	(-1.47)
D_t	1.214	1.991	7.236	7.976	1.305	2.115	7.521	8.100	0.910	1.391	6.078	6.862
	(3.07)	(5.63)	(28.06)	(31.19)	(3.01)	(5.97)	(22.74)	(30.04)	(2.04)	(3.18)	(6.16)	(7.55)
dD_t	-0.338	-0.375	-2.248	-1.871	-0.322	-0.345	-2.129	-1.618	-0.025	-0.084	-1.920	-1.894
	(-1.93)	(-2.32)	(-5.52)	(-4.37)	(-1.35)	(-1.63)	(-4.89)	(-3.62)	(-0.10)	(-0.39)	(-3.21)	(-3.23)
dD_{t+1}	0.659	0.969	2.921	3.664	0.840	1.164	3.318	4.094	0.400	0.615	2.260	2.709
	(5.38)	(9.21)	(9.51)	(10.58)	(7.13)	(12.69)	(11.93)	(12.31)	(2.40)	(3.78)	(3.15)	(3.71)
dV_{t+1}	-0.244	-0.286	-0.125	-0.169	-0.234	-0.281	-0.115	-0.163	-0.304	-0.327	-0.201	-0.252
	(-5.97)	(-6.33)	(-1.31)	(-1.61)	(-5.62)	(-6.04)	(-1.24)	(-1.54)	(-6.78)	(-6.81)	(-1.84)	(-2.04)
C_t	0.661		1.792		0.821		2.098		0.172		0.611	
	(6.11)		(10.05)		(6.69)		(10.09)		(2.39)		(8.89)	
dC_t		0.744		1.238		0.860		1.432		0.288		0.550
		(15.92)		(5.15)		(15.22)		(4.78)		(5.07)		(4.09)
dC_{t+1}		0.962		1.093		1.092		1.225		0.555		0.710
		(8.17)		(4.62)		(8.05)		(4.63)		(8.39)		(5.86)
Const.	0.832	1.026	0.566	0.796	0.866	0.970	0.515	0.821	1.090	1.069	0.648	0.698
	(33.40)	(78.95)	(44.99)	(29.72)	(43.98)	(90.94)	(27.86)	(31.71)	(54.17)	(60.62)	(13.17)	(13.82)
R^2	0.279	0.312	0.376	0.355	0.286	0.323	0.390	0.362	0.305	0.318	0.345	0.354
Ν	48,240	47,807	48,240	47,807	35,295	34,881	35,295	34,881	12,945	12,926	12,945	12,926
Groups	8661	8604	10	10	6404	6348	10	10	2264	2263	10	10

This table shows the estimation results of the reduced valuation regression in Eq. (3). The dependent variable in all specifications is the total market value of the firm (market value of equity plus book value of debt), denoted as V_t . E_t is earnings before interest and extraordinary items (after depreciation and taxes); NA_t is net assets (book value of total assets minus cash); RD_t is research and development (R&D) expenditure; I_t is interest expense; D_t is total dividends paid; and C_t is cash holdings in year t. $dX_t = X_t - X_{t-1}$ denotes the past 1-year change of variable X_t , and $dX_{t+1} = X_{t+1} - X_t$ is the future 1-year change of variable X_t . All variables are scaled by total assets (A_t). We estimate regression includes 10 cross-sections. Year dummy variables are included in all specifications. The data set covers the period from 1995 to 2005. All firms from the different countries are included for which I/B/E/S provides analysts' forecasts and for which company data is available from Worldscope (except financial firms and utilities). The main proxy for information asymmetry, *DISPM*, is based on the standard deviation of analysts' earnings per share forecasts (analysts' forecasts dispersion), and this measure can only be computed when the forecasts are at least based on two analysts. A firm is omitted from the sample if *DISPM* cannot be calculated in at least one sample year. All variables are trimmed at the 1% and the 99% tails. t-values are presented in parentheses. The R^2 of the fixed effects regression refers to the within-dimension. The R^2 of the Fama–MacBeth regression is the average value of the R-squares of the single years.

Significance at the 10% level.

** Significance at the 5% level.

"" Significance at the 1% level.

the control of agency conflicts.¹⁵ In results not reported, we estimate a model where we omit all observations for *DISPM-DUMMY* if our dispersion measure *DISPM* cannot be computed due to lack of data. Presumably, this approach eliminates firms with the highest degree of information asymmetry from the sample. In fact, with this smaller sample of 34,876 firm-year observations the coefficient on the interaction term increases to -0.198, which implies a slightly higher market value of an additional dollar of cash in connection with high information asymmetry (albeit still with a large discount).

In order to control for the direct influence of information asymmetry on firm value, we also include *DISPM*, *DISPM-DUMMY*, and *IA-INDEX* in levels into the corresponding models. The coefficients are significantly negative in most specifications, suggesting that

¹⁵ Based on the median value of *IA-INDEX* (14; Panel A of Table 2), the market value of an additional dollar of cash in Panel C of Table 4 is 0.555 (= 4.475 – 0.280 × 14). An increase in *IA-INDEX* by one standard deviation would even imply a negative market value of an additional dollar of cash. However, the limitations of this interpretation become apparent when the scaling of variables is changed (e.g., using net assets instead of total assets). In results not reported, we find similar results, although the coefficients on C_t and dC_t change considerably in some specifications.

Estimated value of cash in connection with information asymmetry.

	Panel A: IA = DISPM				Panel B: $IA = DISPM - DUMMY$				Panel C: $IA = IA - INDEX$			
	Fixed effec	cts	Fama-Mac	Beth	Fixed effects		Fama-Mac	Beth	Fixed effects		Fama-MacBeth	
	Level	Diff.	Level	Diff.	Level	Diff.	Level	Diff.	Level	Diff.	Level	Diff.
Et	3.694	3.438***	1.876	1.359	2.699	2.529	1.205	0.817	3.991	3.948	2.005	1.776***
	(23.42)	(14.31)	(9.54)	(6.33)	(32.19)	(18.01)	(7.51)	(4.61)	(20.28)	(13.51)	(9.63)	(6.86)
dE_t	-0.228**	-0.284**	0.882	0.846**	-0.045	-0.118	0.923	0.867	-0.206	-0.349**	0.945	0.918
	(-2.29)	(-2.51)	(3.56)	(3.22)	(-0.70)	(-2.41)	(4.92)	(4.72)	(-1.38)	(-2.28)	(2.84)	(2.43)
dE_{t+1}	2.168	1.880	1.938	1.422	1.632	1.437***	1.563	1.227	2.487	2.222***	2.145	1.756
	(19.47)	(22.02)	(10.30)	(8.48)	(33.69)	(27.89)	(10.74)	(10.65)	(32.65)	(31.45)	(10.17)	(8.32)
dNAt	0.332	0.307	0.812	0.695	0.302	0.284	0.721	0.632	0.339	0.339	0.827	0.734
	(12.58)	(15.18)	(8.00)	(7.27)	(13.38)	(14.93)	(8.03)	(7.54)	(13.08)	(14.48)	(8.08)	(6.82)
dNA_{t+1}	0.606	0.673	0.489	0.602	0.571	0.632	0.460	0.548	0.556	0.652	0.443	0.580
	(8.00)	(9.90)	(4.62)	(5.36)	(8.56)	(10.17)	(4.75)	(5.26)	(6.83)	(9.15)	(4.42)	(5.01)
RDt	4.190	3.542	5.398	7.183	3.575	3.371	5.250	6.915	5.193	3.524	7.352	7.613
	(16.21)	(10.05)	(12.41)	(13.11)	(14.97)	(10.25)	(14.38)	(14.46)	(30.63)	(10.21)	(12.73)	(12.39)
dRD,	1.121	1.331	2.572	2.664	1.272	1.184	2.879	2.776	1.193	1.328	2.433	2.659
	(2.42)	(3.06)	(3.01)	(2.56)	(3.87)	(4.54)	(3.23)	(2.88)	(2.53)	(2.68)	(2.35)	(2.45)
dRD_{t+1}	6.110	5.292	8.535	9.365	5.607	4.879	8.609	9.215	6.217	5.103	8.067	9.396
	(8.69)	(12.96)	(9.16)	(10.90)	(13.61)	(25.59)	(9.26)	(12.03)	(9.19)	(10.06)	(8.88)	(9.85)
It	-0.849	-1.447**	1035	-2.402	-0.260	-0.811	1.471	-1.333	-0.825	-1.845**	0.134	-3.342***
	(-1.58)	(-2.40)	(1.80)	(-4.29)	(-1.06)	(-3.33)	(2.41)	(-1.93)	(-1.26)	(-2.09)	(0.22)	(-5.33)
dIr	-1.016	-0.951	-1066	-0.121	-1.005	-0.929	-0.884	-0.195	-1.327**	-0.982	-2.382	-0.409
	(-2.34)	(-2.29)	(-0.84)	(-0.15)	(-2.81)	(-3.01)	(-0.86)	(-0.29)	(-2.43)	(-1.55)	(-1.93)	(-0.48)
dI_{t+1}	-3.138	-4.002	-3.038	-4.984	-2.694	-3.443	-2.643	-4.252	-3.234	-4.362***	-4.509	-6.302***
	(-8.07)	(-11.84)	(-2.94)	(-4.51)	(-10.24)	(-13.52)	(-3.18)	(-5.06)	(-8.80)	(-10.03)	(-4.75)	(-5.16)
D_t	0.428	1.440	7.217	7.939	1.107	1.917	6.841	7.649	-0.083	1.144	6.317	7.766
- 1	(1.00)	(4.16)	(19.90)	(23.66)	(2.84)	(5.52)	(24.48)	(27.90)	(-0.20)	(3.12)	(18.68)	(22.26)
dD,	-0.248	-0.303	-2.381	-1.793	-0.348	-0.375	-2.244	-1.847	-0.264	-0.378	-2.348	-1.875
	(-1.24)	(-1.78)	(-5.29)	(-3.93)	(-1.98)	(-2.34)	(-5.47)	(-4.29)	(-1.30)	(-1.84)	(-5.45)	(-3.78)
dD _{t+1}	0.378	0.829	2.883	3.789	0.594	0.909	2.676	3.393	-0.054	0.570	2.166	3.632
uD (+1	(2.01)	(4.64)	(9.90)	(10.47)	(4.53)	(8.27)	(9.15)	(10.47)	(-0.30)	(3.27)	(6.99)	(8.60)
dV	-0.269	-0.304	-0.149	-0.188	-0.243	-0.285	-0.121	-0.166	-0.253	-0.305	-0.130	-0.180
u , 1+1	(-5.95)	(-6.31)	(-154)	(-1.74)	(-5.93)	(-6.31)	(-1.29)	(-1.61)	(-5.68)	(-6.34)	(-1.48)	(-1.65)
IA.	-0.011	-0.065	-0.026	-0.059	-0.039	-0.062	-0.142	-0.154	-0.070	-0.072	-0.061	-0.043
	(-0.55)	(-5.40)	(-0.87)	(-1.65)	(-3.55)	(-7.40)	(-9.88)	(-9.92)	(-3.85)	(-3.89)	(-1.52)	(-0.89)
C,	0.786	()	2.089	()	0.827	()	2.002	()	4.475	()	7.530	(
-1	(5.44)		(11.16)		(5.64)		(12.27)		(8.32)		(13.32)	
dC,	()	0.978	()	1.902***	()	1.059	()	2.402***	()	1.495	()	5.057
		(1434)		(631)		(11 53)		(876)		(5.86)		(6.20)
dCout		1.031		1 205		0.963		1.083		1 018		1 313
uc _{l+1}		(736)		(4 90)		(8.18)		(4.68)		(6.00)		(4.80)
$C_{\star} \times IA_{\star}$	-0.465***	(7.50)	-0 504***	(1.50)	-0.244***	(0.10)	_0 313 ^{**}	(1.00)	_0 280 ^{***}	(0.00)	-0 397***	(1.00)
$c_l \wedge u_l$	(-3.37)		(-3.56)		(-3.48)		(-2.47)		(-8.48)		(-12.14)	
$dC_{\star} \times IA_{\star}$	(3.37)	-0922***	(5.50)	-2.710^{***}	(3.10)	-0.438	(2.17)	-1 646	(0.10)	-0.051***	(12.14)	-0.267^{***}
		(-7.17)		(-6.10)		(-5.88)		(-8.44)		(-2.84)		(-6.67)
Const	0.801***	1 044	0 557***	0.829	0.865***	1.066	0.675***	0.908	0.969***	1 041	0.576***	0.818
const.	(28.89)	(75.15)	(30.78)	(34 29)	(29.74)	(87.25)	(48 37)	(41 19)	(33 56)	(66.87)	(22 58)	(29.20)
	(20.05)	(13.13)	(30.70)	(34.23)	(23.74)	(07.23)	(10.57)	(41.15)	(33.30)	(00.07)	(22.50)	(23.20)
R^2	0.334	0.366	0.406	0.383	0.281	0.314	0.384	0.366	0.362	0.370	0.471	0.399
Ν	34,876	34,555	34,876	34,555	48,240	47,807	48,240	47,807	29,039	28,769	29,039	28,769
Groups	8661	8589	10	10	8661	8604	10	10	7696	7630	10	10

This table shows the estimation results of the main valuation regression in Eq. (4). The dependent variable in all specifications is the total market value of the firm (market value of equity plus book value of debt), denoted as Vr. Er is earnings before interest and extraordinary items (after depreciation and taxes); NAr is net assets (book value of total assets minus cash); RD_t is research and development (R&D) expenditure; I_t is interest expense; D_t is total dividends paid; and C_t is cash holdings in year t. $dX_t = X_t - X_{t-1}$ denotes the past 1-year change of variable X_t , and $dX_{t+1} = X_{t+1} - X_t$ is the future 1-year change of variable X_t . All variables are scaled by total assets (A_t) . The model also includes interaction terms between cash holdings and three measures for information asymmetry, denoted as IA. DISPM measures the scaled dispersion (standard deviation) of 1-year analysts' earnings per share forecasts provided by I/B/E/S. DISPM-DUMMY is a dummy variable which takes the value of one (high information asymmetry) if a firm exhibits a value of DISPM above its country median in a given year or if the firm is not covered by at least two analysts, and hence we cannot compute DISPM, and zero (low information asymmetry) otherwise. IA-INDEX is a comprehensive index of information asymmetry based on the various dimensions of the concept; it is based on quintile rankings of firm size, R&D expenditure, Tobin's Q, the error in analysts' forecasts, and the number of analysts following the firm in a given year. We estimate regressions using fixed effects and Driscoll and Kraay (1998) standard errors. Alternatively, we also estimate the model using the Fama-MacBeth approach, where each regression includes 10 cross-sections. Year dummy variables are included in all specifications. The data set covers the period from 1995 to 2005. All firms from the different countries are included for which I/B/E/S provides analysts' forecasts and for which company data is available from Worldscope (except financial firms and utilities). The main proxy for information asymmetry, DISPM, is based on the standard deviation of analysts' earnings per share forecasts (analysts' forecasts dispersion), and this measure can only be computed when the forecasts are at least based on two analysts. A firm is omitted from the sample if DISPM cannot be calculated in at least one sample year, i.e., if this firm is not covered by at least two analysts in at least one sample year. All variables are trimmed at the 1% and the 99% tails. t-values are presented in parentheses. The R² of the fixed effects regression refers to the within-dimension. The R^2 of the Fama–MacBeth regression is the average value of the *R*-squares of the single years.

^{*} Significance at the 10% level.

^{**} Significance at the 5% level.

"" Significance at the 1% level.

information asymmetry is generally unfavorable from an investor's perspective. Given that some coefficients are insignificant, an

opposing positive effect may also be at work. A higher divergence of opinions among investors tends to increase the market value of securities as only the most optimistic investors engage in trading (Miller, 1977).¹⁶

To be able to differentiate between our opposing hypotheses, Table 5 contains the results from estimating the valuation regression when the full sample is split into different subsamples. For the sake of brevity, we only report the coefficients that are of direct interest and omit the coefficients on all other model variables. Panels A-D in Table 5 report the results when the sample is split using measures for the quality of corporate governance at the country-level. These models are based on the anti-director rights index, the rule of law index, the corruption index, and a country's law tradition.¹⁷ For the three corporate governance indices the sample is divided into two groups according to higher or lower index values than the median country. A high index value indicates that a country either has higher minority shareholder rights or a stricter enforcement of investors' rights, or more generally, that its corporate governance practices are better. In line with Jensen's (1986) free cash flow hypothesis, the coefficient on cash is higher for firms in common law countries and in countries with higher index values. These results confirm previous findings in Pinkowitz et al. (2006), who document that in countries with poor investor protection a dollar of cash is worth less than in countries with high investor protection. In a broader sense, they are also consistent with the finding in Fresard and Salva (forthcoming) that a cross-listing in the US, a country with high investor protection, tends to increase the valuation of excess cash compared to similar domestic firms. Most important, the negative influence of information asymmetry on the market value of cash tends to be stronger for firms in civil law countries and in countries with lower index values. For each fixed effects model we report the results of a Chow-test. The null hypothesis posits that the coefficients on the interaction term estimated over two groups in a model are equal. Although the χ^2 -test statistic indicates that the coefficients on the interaction term vary significantly only in the subsamples that refer to the law tradition, the differences in all sample splits are large in absolute magnitude. These findings again support Hypothesis 2, implying that the problems in the interplay between cash holdings and information asymmetry are more pronounced for firms that operate in countries with more severe agency problems due to poor corporate governance.

Panel E presents the results from sample splits according to the proportion of closely held shares, which is our firm-level corporate governance measure. We follow Morck et al. (1988) and Opler et al. (1999) and choose three cut-off levels for insider ownership: 0-5%, 5-25%, and 25% or more. Presumably, cash has less value and information asymmetry has a more negative impact on firm value when insider ownership falls into the range between 5% and 25% due to an entrenchment effect that dominates an incentive-alignment effect. However, cash tends to have a higher market value in this subsample. This result may be consistent with McConnell and Servaes (1990), who report a positive relationship between firm value and insider ownership up to a fraction of about 45%. The relationship between the coefficient on the interaction term and closely held shares is hard to interpret. The null hypothesis of the Chow-test that all three coefficients on the interaction term are equal cannot be rejected. If anything, the negative influence of information asymmetry on the marginal value of cash is stronger with low insider shareholdings.

The next two sample splits in Panels F and G in Table 5 are based on country-level financing practices. We use the ratio of both stock and private bond market capitalization to GDP as measures of financial constraints. Presumably, in countries with lower ratios internal finance is of particular importance and cash holdings are highly valuable because firms incur heavy costs of accessing capital markets. A less negative coefficient on the interaction term or even a positive relationship between cash holdings and information asymmetry for firms in constrained countries would support Hypothesis 1, suggesting that cash has a higher market value when information asymmetry is more pronounced. However, we observe the opposite result, where both coefficients on cash and on the interaction term are smaller for firms in countries with lower ratios. One obvious explanation for this result is based on the correlation of a country's financing and corporate governance practices. For example, common law countries are typically marketbased countries, and one would expect that these countries exhibit higher financial development than civil law countries, as indicated by a higher ratio of stock and bond market capitalization to gross domestic product. La Porta et al. (2000b) document that minority shareholders are better protected in common law countries. Similarly, Pinkowitz et al. (2006) suggest that cash is worth less in countries with a low level of financial development because these countries have poor investor protection. Another explanation is based on the role of financial intermediaries and their impact on information asymmetry. Civil law countries tend to be bank-based economies, where financial intermediaries play a major role. Leland and Pyle (1977) suggest that financial intermediaries should be considered as a natural response to information asymmetry. In contrast to shareholders and bondholders, they have privileged access to information and know more about a firm's prospects than minority shareholders. Presumably, the adverse selection problem is less important for banks than for other investors. In marketbased countries, where firms tend to access financial markets to raise funds, information asymmetry could be more pronounced than in bank-based countries. Overall, Hypothesis 1 could be more important for firms in common law countries, which implies a higher (albeit still negative) coefficient on the interaction term in these countries.

Finally, Panel H in Table 5 presents the results from sample splits according to firm size (measured as equity market capitalization) as our firm-level characteristic of financial constraints. Based on the fixed effects model, cash tends to have a higher market value in the group of small firms compared to large firms. This finding is consistent with the notion that large firms are less constrained. Moreover, the coefficients on the interaction term suggest that the negative effect of information asymmetry on the market value of cash is weaker for small firms; the Chow-test rejects the null hypothesis of equal coefficients across groups in the fixed effects model. Accordingly, at least for constrained firms both effects of our conflicting hypotheses seem to be at work. The overall negative effect of information asymmetry on the market value of an additional dollar of cash (Hypothesis 2) is to some extent canceled out by an opposing effect that cash is relatively more valuable in periods with pronounced information asymmetry due to high adverse selection costs (Hypothesis 1). This result is all the more important as firm size is a proxy variable for both financial constraints and information asymmetry, where small firms are more constrained and suffer from higher information asymmetry (see Section 3.1.2).

4.2. Robustness tests

In this section, we undertake a number of robustness tests that provide additional evidence in support of Hypothesis 2, suggesting that liquid assets are valued at a discount with increasing information asymmetry. For the ease of comparison, Panel A of Table 6 presents the base case coefficients on cash holdings and on the

¹⁶ Diether et al. (2002) provide empirical evidence for this model. They argue that prices tend to reflect the views of the most optimistic investors whenever there is a disagreement about a stock's value because the pessimistic investors' views are often not revealed due to short-sale constraints.

¹⁷ The two sample splits based on the rule of law index and the corruption index are identical, and hence they deliver the same estimation results.

Sample splits.

	Fixed effects	Fama-MacBeth	Fixed effects	Fama-MacBeth		Fixed effects	Fama-MacB	eth Fixed effect	s Fama-MacBeth	
	Panel A: Anti-	director rights ind	ector rights index			Panel B: Rule	Panel B: Rule of law index			
	High		Low			High		Low		
C_t $C_t \times DISPM_t$ N Groups Chow-test	1.059 ^{***} (6.40) -0.243 (-1.56) 19,958 4844	$\begin{array}{c} 2.435^{\ast\ast\ast} \\ (11.11) \\ -0.310 \\ (-1.55) \\ 19.958 \\ 10 \\ \chi^2(1) \end{array}$	$0.196 \\ (1.63) \\ -0.683 \\ (-5.23) \\ 14,006 \\ 3573 \\ = 1.44$	1.246 ^{***} (5.13) -0.532 ^{**} (-2.37) 14,006 10	C_t $C_t \times DISPM_t$ N Groups Chow-test	0.883 ^{***} (5.87) -0.391 ^{**} (-2.53) 26,831 6304	2.390 ^{***} (11.08) -0.404 ^{**} (-2.38) 26,831 10		0.910 ^{***} (5.31) -1.197 ^{**} (-3.22) 8045 10	
	Panel C: Corru	uption index				Panel D: Law	tradition			
	High		Low			Common law		Civil law		
C_t $C_t \times DISPM_t$ N Groups Chow-test	0.883*** (5.87) -0.391** (-2.53) 26,831 6304 Panel E: Perce	2.390 ^{***} (11.08) -0.404 ^{**} (-2.38) 26,831 10 $\chi^2(1)$ entage of closely he	0.376^{**} (2.77) -0.990^{***} (-8.21) 8045 2364 = 1.74 eld shares	0.910 ^{***} (5.31) -1.197 [*] (-3.22) 8045 10	C_t $C_t \times DISPM_t$ N Groups Chow-test	$\begin{array}{c} 1.063^{***} \\ (5.92) \\ -0.158 \\ (-0.88) \\ 18,510 \\ 4565 \end{array}$	2.438 ^{***} (10.72) -0.269 (-1.04) 18,510 10	$\begin{array}{c} 0.225^{***} \\ (2.30) \\ -0.757^{***} \\ (-6.95) \\ 16,366 \\ 4109 \\ ^2(1) = 3.05^{*} \end{array}$	1.281^{***} (5.98) -0.493^{*} (-2.02) 16,366 10	
	< 5%		5-25%		> 25%					
C_t $C_t \times DISPM_t$ N Groups Chow-test	0.968 ^{***} (7.45) -1.097 ^{***} (-2.86) 3502 1236	2.288 ^{***} (6.04) 0.764 (0.63) 3502 10	$\begin{array}{c} 1.079^{***} \\ (4.12) \\ -0.378^{*} \\ (-1.78) \\ 7366 \\ 2862 \\ \chi^2(2) = \end{array}$	2.586^{+++} (9.69) -0.183 (-0.68) 7366 10 = 1.79	0.310 ^{***} (3.02) -0.137 (-0.73) 17,490 5585	1.938*** (11.64) -0.884 (-4.62) 17,490 10				
	Panel F: Stock	x/GDP				Panel G: Bond	/GDP			
	High		Low			High		Low		
C_t $C_t \times DISPM_t$ N Groups Chow-test	0.844*** (5.31) -0.317* (-2.04) 28,788 6859	$\begin{array}{c} 2.319 \\ (11.24) \\ -0.332 \\ (-2.00) \\ 28,788 \\ 10 \\ \chi^2(1) \end{array}$	$\begin{array}{c} 0.373^{\bullet\bullet\bullet}\\ (6.18)\\ -1.040^{\bullet\bullet}\\ (-6.79)\\ 6088\\ 1810\\ = 3.22^{\bullet}\end{array}$	0.956 ^{***} (9.60) -0.991 (-3.77) 6088 10	C_t $C_t \times DISPM_t$ N Groups Chow-test	0.817*** (4.80) -0.375* (-2.51) 26,045 6400	2.395 ^{***} (11.57) -0.556 ^{***} (-4.57) 26,045 10 <i>X</i>	$\begin{array}{c} 0.649^{***} \\ (3.71) \\ -0.634^{*} \\ (-2.22) \\ 8259 \\ 2071 \\ ^2(1) = 0.28 \end{array}$	1.066^{***} (6.34) -1.047 (-1.69) 8259 10	
	Panel H: Firm	size								
	Large		Small							
C_t $C_t \times DISPM_t$ N Groups Chow-test	0.547*** (-3.73) -1.252*** (-6.32) 17,435 4397	$\begin{array}{c} 2.235^{\ast\ast\ast} \\ (-9.58) \\ -0.729 \\ (-1.07) \\ 17,435 \\ 10 \\ \chi^2(1) = \end{array}$	0.789 ^{***} (-5.10) -0.135 (-0.90) 17,441 5438 = 7.03 ^{***}	$\begin{array}{c} 1.845^{***} \\ (-9.44) \\ -0.435 \\ (-1.59) \\ 17,441 \\ 10 \end{array}$						

This table shows the estimation results of the main valuation regression in Eq. (4) for subsamples when the full sample is split using measures for the quality of corporate governance and financial constraints. The dependent variable in all specifications is the total market value of the firm (market value of equity plus book value of debt), denoted as Vt. For the sake of brevity, the table only reports the coefficients that are of direct interest. Ct is cash holdings in year t, and DISPM in the interaction term with cash measures the scaled dispersion (standard deviation) of 1-year analysts' earnings per share forecasts provided by I/B/E/S. We estimate regressions using fixed effects and Driscoll and Kraay (1998) standard errors. Alternatively, we also estimate the model using the Fama-MacBeth approach, where each regression includes 10 cross-sections. Year dummy variables are included in all specifications. Panels A-D show the results when the sample is split based on the anti-director rights index, the rule of law index, the corruption index, and a country's law tradition. For the three governance indices, the sample is divided into two groups according to higher or lower index values than the median country using data for the year 2000. A high index value indicates that a country either has higher minority shareholder rights or stricter enforcement of investors' rights, or more generally, that its corporate governance practices are better. Panel E splits the sample based on the percentage of insider ownership, which is taken as a proxy for firm-level corporate governance. The sample splits in Panels F and G are based on financing practices at the country-level. The ratios of both stock and private bond market capitalization to GDP (using data for the year 2000) are measures for financial constraints. Firms in countries with lower ratios tend to be constrained. Finally, in Panel H the sample is split based on firm size, measured as equity market capitalization, which is taken as a proxy for financial constraints on the firm-level. The data set covers the period from 1995 to 2005. All firms from the different countries are included for which I/B/E/S provides analysts' forecasts and for which company data is available from Worldscope (except financial firms and utilities). The main proxy for information asymmetry, DISPM, is based on the standard deviation of analysts' earnings per share forecasts (analysts' forecasts dispersion), and this measure can only be computed when the forecasts are at least based on two analysts. A firm is omitted from the sample if DISPM cannot be calculated in at least one sample year, i.e., if this firm is not covered by at least two analysts in at least one sample year. All variables are trimmed at the 1% and the 99% tails. tvalues are presented in parentheses. For each fixed effects model the result of a Chow-test is reported. The null hypothesis is that the coefficients on the interaction term estimated over the groups in a model are equal. The test statistic is χ^2 -distributed, with the number of degrees of freedom shown in brackets.

Significance at the 10% level.

Significance at the 5% level.
 Significance at the 1% level.

interaction variable between cash and our main proxy variable for information asymmetry (DISPM), as they have been shown in Table 4 for the full sample. Again, for the sake of brevity, we only report the coefficients that are of direct interest. Moreover, we focus on the results of the fixed effects model including the level of cash. In addition to the results for the full sample, we follow the setup in Table 3 and report the results for the subsamples of firms from developed and emerging markets. As one would expect, the coefficient on the interaction term is larger in absolute value for firms in emerging markets than in developed markets. While the negative coefficients generally support Hypothesis 2, the valuation discount on an additional dollar of cash with increasing information asymmetry is more pronounced in emerging markets. These findings extend the result in Pinkowitz et al. (2006) that cash is worth less in countries with a low level of financial development because these countries have poor investor protection. They are in line with our own results on the relationship between the market value of cash and information asymmetry when the sample is split according to

the quality of corporate governance (see Table 5). The sample splits on a country-level allow us to disentangle the effects of our two conflicting hypotheses. In Table 5 we use median-splits (where applicable) based on data for the year 2000, which is the year in the middle of our sample period. Kaufmann et al. (2008) document that while the changes of corporate governance in most countries were small over the last decade, some countries exhibited significant changes. To check the robustness of our results, we split the sample using the data from Kaufmann et al. (2008) for the rule of law index and the corruption index for 1998 and 2005.¹⁸ There is no time series for the anti-director rights index. We also use the relative sizes of the stock and bond markets for 1995 and 2005 in these robustness tests. In results not reported, our findings remain robust irrespective of whether we use data from the beginning, the middle, or the end of our sample period for the splits. While some countries exhibit changes in the index levels, the relative ranking of countries and hence the regression results often do not change.

In another robustness test, we make three changes to the specification of our main model. The results are shown in Panels B-D of Table 6. Our main regression is based on the valuation model of Fama and French (1998). While they use 2-year changes for the explanatory variables in differences to model investors' expectations in their original model, we follow Pinkowitz et al. (2006) and Dittmar and Mahrt-Smith (2007) and only use 1-year changes. As shown in Panel B, using 2-year changes reduces our sample, but we still observe a negative influence of information asymmetry on the market value of cash holdings. Moreover, Panel C shows that the coefficients and their statistical inference do not change qualitatively when we estimate the valuation regressions omitting time dummy variables. Finally, in Panel D we estimate the model using ordinary least squares with cluster robust standard errors rather than fixed effects. While the coefficient on the level of cash changes considerably, that on the interaction term is more stable and remains significant.

To compute our main measure of information asymmetry, *DIS-PM*, we use the average of monthly dispersions in each year. If we took the data only for one specific month, we lose all firm-year observations for which no (updated) estimate for this particular month is available. Towards the end of a year dispersion decreases because unexpected events become less probable and uncertainty will be resolved. As we cannot measure dispersion for each firm in every month, this monthly average may underestimate the dispersion of firms for which we have no observations in the first months of the year. In fact, forecasts are only available for a small propor-

|--|

Robustness	tests.
------------	--------

	All countries	Developed markets	Emerging markets
Panel A (base case)			
Ct	0.786	0.893	0.276
	(5.44)	(5.63)	(2.54)
$C_t \times DISPM_t$	-0.465	-0.363**	-1.170
	(-3.37)	(-2.39)	(-7.65)
Ν	34,876	27,859	7017
Groups	8661	6551	2116
Panel B (2-year lags)			
Ct	0.440	0.551	-0.212
	(1.98)	(2.42)	(-1.09)
$C_t \times DISPM_t$	-0.700	-0.689	-0.661
	(-3.35)	(-3.08)	(-3.35)
N	24,820	20,125	4695
Groups	6576	5041	1539
Panel C (no time dumn	nies)		
C_t	0.841	0.931	0.365
	(5.43)	(5.60)	(2.76)
$C_t \times DISPM_t$	-0.458	-0.314	-1.335
N	(-3.01)	(-1.92)	(-6.24) 7017
N Croups	34,870 9661	27,809 GEE1	7017
Groups	8001	0331	2110
Panel D (pooled OLS)			
C_t	2.215	2.494	1.042
	(20.35)	(21.03)	(3.90)
$C_t \times DISPINI_t$	-0.007	-0.595	-1.101
N	(-2.75)	(-2.51)	(-2.51) 7017
Groups	8661	6551	2116
Danal F (valatility)	0001	0001	2110
Panel E (Volatility)	0.562***	0.681	0 222
C _t	(2.79)	(3.23)	(-0.77)
$C_{t} \times DISPM_{t}$	-0.533	-0.426	-1 413
	(-3.94)	(-2.85)	(-9.03)
VOLAt	-0.238	-0.335	-0.128
	(-1.71)	(-2.57)	(-0.79)
$C_t \times VOLA_t$	1.548	1.480	3.862
	(1.82)	(1.69)	(2.38)
Ν	34,383	27,466	6917
Groups	8579	6489	2096
Panel F (excess cash)			
EXCASH _t	2.141	2.113	3.023
	(7.60)	(7.35)	(4.65)
$EXCASH_t \times DISPM_t$	-0.513	-0.488	-1.235
	(-1.81)	(-1.70)	(-1.27)
N	12,900	11,195	705
Groups	3564	3315	249
Panel G (payout ratio)			
Dt	1.196	1.347	0.296
	(2.42)	(2.87)	(0.39)
$POR_t \times DISPM_t$	0.139	0.172	0.296
N	(7.93)	(6.18)	(0.39)
IN Crowne	54,598	27,639	2112
Groups	8646	0540	2112

This table shows results of different robustness tests based on the main valuation regression in Eq. (4). The dependent variable in all specifications is the total market value of the firm (market value of equity plus book value of debt), denoted as V_t . For the sake of brevity, the table only shows the coefficients that are of direct interest. Ct is cash holdings in year t, and DISPM in the interaction term with cash measures the scaled dispersion (standard deviation) of 1-year analysts' earnings per share forecasts provided by I/B/E/S. Moreover, only the results of the fixed effects model including the level of cash are reported. Panel A reports the results of the base model specification. Changes in the model specification involve using 2-year changes for the explanatory variables in differences (Panel B), omitting time dummy variables (Panel C), and using ordinary least squares with cluster robust standard errors rather than fixed effects (Panel D). The model in Panel E incorporates the standard deviation of monthly stock returns over the accounting year as a direct measure of risk as well as the interaction term between cash and stock return volatility. Panel F reports the results from the alternative specification of the valuation regression in Eq. (6). This model uses excess cash instead of cash, where excess cash is computed as the difference between actual cash holdings and the target cash level in Eq. (5). Only positive values of excess cash are included. Panel G shows a variant of the valuation regression, where DISPM is interacted with the payout ratio (POR), defined as the ratio of total dividends and share repurchases to operating income, instead of cash. The data set covers the period from 1995 to 2005. All firms from the different countries are included for which I/B/E/S provides analysts' forecasts and for which company data is available from Worldscope (except financial firms and utilities). The main proxy for information asymmetry is based on the standard deviation of analysts' earnings per share forecasts (analysts' forecasts dispersion), and this measure can only be computed when the forecasts are at least based on two analysts. A firm is omitted from the sample if this dispersion measure cannot be calculated in at least one sample year, i.e., if this firm is not covered by at least two analysts in at least one sample year. All variables are trimmed at the 1% and the 99% tails. t-values are presented in parentheses.

Significance at the 10% level.

Significance at the 5% level.

Significance at the 1% level.

¹⁸ The data are available at http://www.govindicators.org.

tion of our sample firms during the first quarter, and dispersion varies widely. Therefore, as an additional robustness check (not reported), we compute the average dispersion in each quarter. When we use the average dispersion in the second, third, and fourth quarter instead of all available months, our main results do not change qualitatively. However, when we use the average dispersion in the first quarter, the interaction term (albeit still negative) becomes insignificant, probably indicating that these far-distant forecasts have little predictive value (Capstaff et al., 2001).

Another concern is a potential correlation between risk (in a broader sense) and information asymmetry. In the spirit of Diether et al. (2002), we include two additional variables into the valuation regression. First, we add the standard deviation of monthly stock returns over the accounting year as a direct measure of risk. Second, we include the interaction term between cash and stock return volatility. As documented in Panel E of Table 6, we find a significantly negative coefficient on the interaction term between cash and information asymmetry. In contrast, the estimated coefficient on the interaction term between cash and risk is significantly positive. This result can be explained by the notion that cash is more valuable when a firm's business risk is higher. Most important, the influences of risk and information asymmetry run into opposite directions, and hence we conclude that our results cannot be explained by a positive correlation between our measure of information asymmetry and risk.

The alternative valuation regression in Eq. (6) proposed by Dittmar and Mahrt-Smith (2007) includes a measure of excess cash, as specified in the model in Eq. (5). Panel F of Table 6 presents the results when only positive values of excess cash (EXCASH) are included. The coefficient on EXCASH is statistically significant. However, it is as large as 2.141, which implies that one dollar put into excess cash increases firm value by more than double its par value. The finding that the coefficient on excess cash is much larger than that on cash in our main model in Table 4 is clearly surprising. However, it is comparable in magnitude to that in Dittmar and Mahrt-Smith (2007), who report that a dollar of excess cash increases firm value by two to three dollars, depending on the corporate governance measure they use. An immediate explanation is the endogeneity between excess cash and firm value, which may lead to biased coefficients. The market-to-book ratio, as a proxy for investment opportunities, determines cash holdings. However, cash holdings themselves affect the market value of the firm and the market-to-book ratio. As in Dittmar and Mahrt-Smith (2007), we use an instrumented market-to-book ratio to compute the normal cash level. However, they only focus on the interpretation of the interaction term.¹⁹

Most important, our results in Panel F reveal that the coefficient on the interaction term is significantly negative. More pronounced information asymmetry decreases the benefit of holding excess cash. This finding again corroborates Hypothesis 2, which is related to the free cash flow theory. In order to illustrate the detrimental value effect of information asymmetry, we again calculate the market value of an additional dollar of excess cash in connection with information asymmetry. The coefficient on excess cash is 2.141. If information asymmetry is taken into account, the marginal value of excess cash reduces slightly to 2.084 (based on the median value of DISPM of 0.112 in Panel A of Table 2). Increasing information asymmetry by one standard deviation (0.261; Panel A of Table 2), the market value of one additional dollar of excess cash decreases by 0.134 (or 6.9%) to 1.950. Overall, we again conclude that the market value of excess cash is both statistically and economically lower in states with high information asymmetry. According to

Hypothesis 2, the agency costs from hoarding liquidity dominate the potential savings from the availability of internal funds when the degree of information asymmetry is high. As in our main model in Panel A, the decrease in the market value of an additional dollar of cash with increasing information asymmetry is more pronounced in emerging markets than in developed markets. However, the coefficient on the interaction term is estimated insignificantly for the emerging markets' subsample, which may be attributable to the small number of firm-year observations. In oder to capture country-specific effects, the target cash level is estimated separately for each country (Fresard and Salva, forthcoming). Due to small sample sizes for some of the emerging markets, the estimation of the target cash level (and hence excess cash) may be imprecise. When we estimate the coefficients for the target cash level in Eq. (5) using all emerging markets' observations together, the coefficient on the interaction term in Eq. (6) for this subsample becomes significantly negative at the 5% level. This reinforces the findings from our main model in Panel A.²⁰

In a final robustness test, we examine the contribution of payouts to firm value in connection with information asymmetry. Cash and dividend decisions are closely related. Private benefits create a wedge between the value of a dollar inside the firm and the value of a dollar paid out. No private benefits can be consumed from a dollar paid out, while a dollar kept within the firm potentially induces insiders to consume private benefits. As argued by Pinkowitz et al. (2006), if investors discount the value of cash holdings because they expect the cash to be partly wasted in perquisites, they should value dividends in that country at a premium compared to a country where private benefits are less important. With stickiness in dividends, high current dividends predict high future dividends and a lower consumption of private benefits. Similarly, La Porta et al. (2000a) document that firms experience more pressure to pay dividends in countries with poor investor protection because firms' resources are more likely to be consumed as private benefits. If investor protection is sufficiently weak, limiting private benefits through dividend payments will more than offset a tax disadvantage of dividends. Using the reduced valuation model in Eq. (3). Pinkowitz et al. (2006) document that dividends contribute more to firm value in countries with poorer investor protection.²¹ While we also report significantly positive coefficients on D_t in Table 3, we cannot find that the coefficients are higher in emerging markets with poor investor protection and low financial development. More important, Panel G of Table 6 presents the results of our main valuation regression in Eq. (4) when DISPM is interacted with the payout ratio, denoted as POR, instead of cash. Dividends are paid out from earnings rather than from assets, and hence POR is computed as the ratio of total dividends and share repurchases to operating income. For the full sample the coefficient on the interaction term is significantly positive, implying that the market value of an additional dollar paid out from earnings increases when information asymmetry is high. We interpret this finding as indirect support for Hypothesis 2, indicating that in states with a higher degree of information asymmetry cash is less valuable if kept inside the firm rather than being paid out to shareholders. However, the hypothesis that this effect is more pronounced in emerging markets than in developed markets is not confirmed by our data, as shown by the coefficients on the interaction term in the subsamples.

¹⁹ In order to avoid an omitted variables bias from excluding the target cash level, we test alternative models that include (C - EXCASH) and (C - EXCASH) × DISPM. However, the coefficient on excess cash remains very high.

²⁰ In other results not reported, these findings are even more pronounced when we estimate the target cash level using all sample firms rather than on a country-specific basis or for a group of countries (i.e., developed and emerging markets).

²¹ In a related analysis, Ferris et al. (2009) show that when the legal regime and the level of investor protection permit, investors force dividends from managers, but they also extract such payouts indirectly by placing a higher value on dividend paying firms.

5. Conclusions

This study examines the value effects of corporate cash holdings in a novel setting. Previous literature on cash holdings explores the valuation effects by differentiating firms along their quality of corporate governance. We take a different perspective and focus on the valuation effects of cash in connection with firm-specific and time-varying information asymmetry. Specifically, we test two conflicting hypotheses. First, based on Myers and Majluf (1984), cash holdings in combination with a higher level of information asymmetry have a positive influence on firm value because the adverse selection costs that arise from external finance can be avoided. Second, Jensen's (1986) free cash flow theory coupled with a higher level of information asymmetry leads to extreme moral hazard. If this effect dominates, the market value of cash incurs a discount with more severe information asymmetry. the University of Vienna, the 2007 CFS Summer School in Eltville, the 2009 Midwest Finance Association (MFA) Meeting in Chicago, the 2009 Financial Management Association (FMA) Meeting in Turin, and the 2009 European Financial Management Association (EFMA) Meeting in Milan for helpful comments. We thank Rebekka Haller for her excellent research assistance. The views and opinions expressed herein are those of the author and do not necessarily represent the views and opinions of KPMG LLP (UK). The information contained is of a general nature and is not intended to address the circumstances of any particular individual or entity.

Appendix

The detailed formula for our main measure of information asymmetry ($DISPM_{it}$) is:

$$DISPM_{i,t} = \ln \left(1 + \frac{1}{M_{i,t}} \times \sum_{m_{i,t}=1}^{M_{i,t}} \sqrt{\left(\frac{\frac{1}{A_{m_{i,t}}-1} \times \sum_{a_{m_{i,t}}=1}^{A_{m_{i,t}}} \left(EPS_{a_{m_{i,t}}} - \frac{1}{A_{m_{i,t}}} \times \sum_{a_{m_{i,t}}=1}^{A_{m_{i,t}}} EPS_{a_{m_{i,t}}} \right)^2}{Med_{m_{i,t}}} \right) \right),$$

In order to examine these two opposing hypotheses, we examine a data set covering more than 8500 firms from 45 countries over the period from 1995 to 2005. We use an extended version of the valuation regression of Fama and French (1998) and take the dispersion of analysts' earnings forecasts as our main measure of firm-specific and time-varying information asymmetry. Our results indicate that the market value of one dollar is on average around one (albeit with a wide variation across models). Most important, the market value of cash is significantly reduced when a firm faces a higher level of information asymmetry. This evidence suggests that the agency costs of free cash flow outweigh the benefits of cash as an internal source of finance. To further distinguish between the two opposing hypotheses, we split the sample according to the quality of corporate governance and financial constraints. These sample splits reinforce our finding that agency costs due to moral hazard decrease the market value of an additional dollar of cash. Given high information asymmetry, the market value of cash is higher if investor protection is better and the quality of corporate governance is higher. In contrast, the hypothesis that cash is valued higher if a firm is financially constrained is only partly confirmed.

In summary, our results indicate that the agency costs based on the free cash flow theory outweigh the benefits of financial slack in mitigating adverse selection costs when raising external funds. Put differently, it may not be in the shareholders' interest that firms hoard liquidity due to problems induced by higher levels of information asymmetry. The precautionary motive to keep funds within the firm seems questionable. However, our findings do not contradict the pecking order theory in general. We do not suggest that firms should not use internal funds in the first place before external funds are raised. Instead, we argue that it may not be optimal for firms to accumulate cash rather than pay it out with the intention to avoid external finance in future states when information asymmetry is high.

Acknowledgements

We thank an anonymous referee, Marco Becht, Alice Bonaime, Francesca Cornelli, Daniel Höchle, Ben Jann, Ike Mathur (the editor), David Yermack, Heinz Zimmermann, Josef Zechner, and participants of workshops at the University of Basel, the University of Innsbruck, where $Med_{m_{i,t}}$ is the absolute median earnings per share forecast in month *m* in year *t* for firm *i*; $A_{m_{i,t}}$ is the number of analysts that cover firm *i* in year *t* in month *m*; $M_{i,t}$ is the number of months during which more than two analysts cover firm *i* in year *t*; and the $EPS_{a_{m_{i,t}}}$ is earnings per share estimate of analyst a for firm *i* in year *t* in month *m*.

References

- Aboody, D., Lev, B., 2000. Information asymmetry, R&D, and insider gains. Journal of Finance 55, 2747–2766.
- Almeida, H., Campello, M., Weisbach, M.S., 2004. The cash flow sensitivity of cash. Journal of Finance 59, 1777–1804.
- Autore, D., Kovacs, T., 2010. Equity issues and temporal variation in information asymmetry. Journal of Banking and Finance 34, 12–23.
- Bates, T.W., Kahle, K.M., Stulz, R.M., 2009. Why do US firms hold so much more cash than they used to? Journal of Finance 74, 1985–2021.
- Bessler, W., Stanzel, M., 2009. Conflicts of interest and research quality of affiliated analysts in the german universal banking system: Evidence from IPO underpricing. European Financial Management 15, 590–620.
- Bharat, S., Pasquariello, P., Wu, G., 2008. Does asymmetric information drive capital structure decisions. Review of Financial Studies 22, 3211–3243.
- Brennan, M., Subrahmanyam, A., 1995. Investment analysis and price formation in securities markets. Journal of Financial Economics 38, 361–381.
- Cai, J., Liu, Y., Qian, Y., 2009. Information asymmetry and corporate governance. Working Paper, Drexel University.
- Capstaff, J., Paudyal, K., Rees, W., 2001. A comparative analysis of earnings forecasts in Europe. Journal of Business Finance and Accounting 28, 531–562.
- Chang, X., Dasgupta, S., Hillary, G., 2006. Analysts coverage and financing decisions. Journal of Finance 61, 3009–3048.
- Choe, H., Masulis, R.W., Nanda, V.K., 1993. Common stock offerings across the business cycle: Theory and evidence. Journal of Empirical Finance 1, 3–31.
- Diamond, D., Verrecchia, R., 1991. Disclosure, liquidity, and the cost of capital. Journal of Finance 46, 1325–1359.
- Diether, K.B., Malloy, C.J., Scherbina, A., 2002. Differences of opinion and the cross section of stock returns. Journal of Finance 57, 2113–2141.
- Dittmar, A., Mahrt-Smith, J., 2007. Corporate governance and the value of cash holdings. Journal of Financial Economics 83, 599–634.
- Dittmar, A., Mahrt-Smith, J., Servaes, H., 2003. International corporate governance and corporate cash holdings. Journal of Financial and Quantitative Analysis 38, 111–133.
- D'Mello, R., Ferris, S., 2000. The information effects of analyst activity as the announcement of new equity issues. Financial Management 29, 78–95.
- D'Mello, R., Krishnaswami, S., Larkin, P., 2008. Determinants of corporate cash holdings: Evidence from spin-offs. Journal of Banking and Finance 32, 1209–1220.
- D'Mello, R., Miranda, M., 2010. Long-term debt and overinvestment agency problems. Journal of Banking and Finance 34, 324–335.
- Driscoll, J.C., Kraay, A.C., 1998. Consistent covariance matrix estimation with spatially dependent panel data. Review of Economics and Statistics 80, 549– 560.

- Elton, E.J., Gruber, M.J., Gultekin, M.N., 1984. Professional expectations: Accuracy and diagnosis of errors. Journal of Financial and Quantitative Analysis 19, 351– 363.
- Fama, E.F., French, K.R., 1998. Taxes, financing decisions, and firm value. Journal of Finance 53, 819–843.
- Fama, E.F., French, K.R., 2005. Financing decisions: Who issues stock? Journal of Financial Economics 76, 549–582.
- Fama, E.F., Jensen, M.C., 1983. Separation of ownership and control. Journal of Law and Economics 26, 301–325.
- Faulkeneder, M., Wang, R., 2006. Corporate financial policy and the value of cash. Journal of Finance 61, 1957–1990.
- Ferris, S.P., Jayaraman, N., Sabherwal, S., 2009. Catering effects in corporate dividend policy: The international evidence. Journal of Banking and Finance 33, 1730– 1738.
- Fresard, L., Salva, C., forthcoming. The value of excess cash and corporate governance: Evidence from US cross-listings. Journal of Financial Economics.
- Harford, J., Mansi, S., Maxwell, W., 2008. Corporate governance and firm cash holdings. Journal of Financial Economics 87, 535–555.
- Höchle, D., 2007. Robust standard errors for panel data sets with cross-sectional dependence. Stata Journal 7, 281–312.
- Jensen, M.C., 1986. Agency costs of free cash flow, corporate finance, and takeovers. American Economic Review 76, 323–330.
- Jensen, M.C., Meckling, W.H., 1976. Theory of the firm: Managerial behavior, agency costs and ownership structure. Journal of Financial Economics 3, 305– 360.
- Kalcheva, I., Lins, K.V., 2007. International evidence on cash holdings and expected managerial agency problems. Review of Financial Studies 20, 1087–1112.
- Kaufmann, D., Kraay, A., Mastruzzi, M., 2008. Governance matters VII: Aggregate and individual governance indicators 1996–2007. Working Paper, The World Bank.
- Korajczyk, R.A., Lucas, D.J., McDonald, R.L., 1992. Equity issues with time-varying asymmetric information. Journal of Financial and Quantitative Analysis 27, 397–417.
- Krishnaswami, S., Subramaniam, V., 1999. Information asymmetry, valuation, and the corporate spin-off decision. Journal of Financial Economics 53, 73–112.
- Krishnaswami, S., Yaman, D., 2007. Contracting costs and the window of opportunity for straight debt issue. Journal of Banking and Finance 31, 869–888.
- La Porta, R., Lopez-de Silanes, F., Shleifer, A., Vishny, R.W., 1998. Law and finance. Journal of Political Economy 106, 1113–1155.
- La Porta, R., Lopez-de Silanes, F., Shleifer, A., Vishny, R.W., 2000a. Agency problems and dividend policy around the world. Journal of Finance 55, 1–33.
- La Porta, R., Lopez-de Silanes, F., Shleifer, A., Vishny, R.W., 2000b. Investor protection and corporate governance. Journal of Financial Economics 58, 3–27.

- Leary, M.T., Roberts, M.R., 2007. The pecking order, debt capacity, and information asymmetry. Working Paper, Duke University.
- Leland, H.E., Pyle, D.H., 1977. Informational asymmetries, financial structure, and financial intermediation. Journal of Finance 32, 371–387.
- Lundstrum, L., 2003. Firm value, information problems and the internal capital market. Review of Quantitative Finance and Accounting 21, 141–156.
- McConnell, J.J., Servaes, H., 1990. Additional evidence on equity ownership and equity value. Journal of Financial Economics 27, 595–612.
- McLaughlin, R., Safieddine, A., Vasudevan, G., 1998. The information content of corporate offerings of seasoned securities: An empirical analysis. Financial Management 27, 31–45.
- Miller, E.M., 1977. Risk, uncertainty, and divergence of opinion. Journal of Finance 32, 1151–1168.
- Morck, R., Shleifer, A., Vishny, R., 1988. Management ownership and market valuation: An empirical analysis. Journal of Financial Economics 20, 293–315.
- Myers, S.C., 1984. The capital structure puzzle. Journal of Finance 39, 575-592.
- Myers, S.C., Majluf, N.S., 1984. Corporate financing and investment decisions when firms have information that investors do not have. Journal of Financial Economics 13, 187–221.
- Nohel, T., Tarhan, V., 1998. Share repurchases and firm performance: New evidence on the agency costs of free cash flow. Journal of Financial Economics 49, 187–222.
- Opler, T., Pinkowitz, L., Stulz, R., Williamson, R., 1999. The determinants and implications of corporate cash holdings. Journal of Financial Economics 52, 3– 46.
- Ozkan, A., Ozkan, N., 2004. Corporate cash holdings: An empirical investigation of UK companies. Journal of Banking and Finance 28, 2103–2134.
- Petersen, M.A., 2009. Estimating standard errors in finance panel data sets: Comparing approaches. Review of Financial Studies 22, 435–480.
- Peyer, U., Vermaelen, T., 2009. The nature and persistence of buyback anomalies. Review of Financial Studies 22, 1693–1745.
- Pinkowitz, L., Stulz, R., Williamson, R., 2006. Does the contribution of corporate cash holdings and dividends to firm value depend on governance? A cross-country analysis. Journal of Finance 61, 2725–2751.
- Pinkowitz, L., Williamson, R., 2004. What is a dollar worth? The market value of cash holdings. Working Paper, Georgetown University.
- Smith, C., Watts, R., 1992. The investment opportunity set and corporate financing, dividend, and compensation policies. Journal of Financial Economics 32, 263– 292.
- Vermaelen, T., 1981. Common stock repurchases and market signaling: An empirical study. Journal of Financial Economics 9, 139–183.
- Viswanath, P.V., 1993. Strategic considerations, the pecking order hypothesis, and market reactions to equity financing. Journal of Financial and Quantitative Analysis 28, 213–234.